



Article Do Forwarders Improve Sustainability Efficiency? Evidence from a European DEA Malmquist Index Calculation

Matthias Klumpp 1,2

- ¹ Department of Industrial Engineering and Business Information Systems (IEBIS), University of Twente, Drienerlolaan 5, 7522 NB Enschede, The Netherlands
- ² Institute for Logistics and Service Management, FOM University of Applied Sciences, Leimkugelstr. 6, 45141 Essen, Germany; matthias.klumpp@fom.de; Tel.: +49-201-8100-4554

Academic Editor: Ernest Reig-Martínez Received: 27 March 2017; Accepted: 11 May 2017; Published: 18 May 2017

Abstract: Sustainability performance and efficiency is an important topic in transportation and for forwarders. This is shown, for example, by the fact that major logistics service providers LSP publish sustainability reports, often within the annual legal business report. However, in depth research is missing regarding the efficiency of forwarders regarding the established triple bottom line approach for sustainability, including economic, social, and ecology performance areas. This is especially true for a dynamic time-series perspective, as usually only static analyses for one point in time are presented (in most cases single business years). Therefore, the operations research technique of a data envelopment analysis (DEA) Malmquist index calculation is used in order to provide a longitudinal calculation of efficiency, incorporating multiple objectives regarding the triple bottom line approach for European forwarders. Several indicators are tested, including total revenues and assets as input types, profit (EBIT) and dividend volume (economic dimension), employment and gender equality in management (social), and carbon-equivalent emissions (environmental) as output types.

Keywords: sustainability efficiency; DEA Malmquist index; European logistics service providers

1. Introduction

Sustainability is a very germane topic for logistics service providers (LSP) as outlined for example by Abbasi and Nilsson [1] (p. 263), Govindan and Soleimani [2] (p. 371), or Colicchia et al. [3] (p. 189). True for all transport modes in transportation, e.g., for road, rail [4] (p. 148), air, and sea [5] (p. 473) transport [6] (p. 414), an important characteristic of sustainability concepts and efforts with LSP is also the fact that they steer the transport chain on behalf of industry and retail corporations and therefore influence their own sustainability performance [7] (p. 589). Furthermore, this is also proven by the fact that all major LSP provide extensive sustainability data and reports, often within legal business report documents, for example Deutsche Post DHL [8], Kühne + Nagel [9], or A.P. Moeller Maersk [10], and research indicates that these reports drive change [11] as well as firm value [12,13]. This importance also extends towards the question of external institutional legitimation by business partners, as outlined by Busse, Kach, and Bode [14] (p. 321). Although supply management is deemed to play an important role in implementing sustainability objectives [15] and technology advances may help in the endeavor [16,17], some authors search for a "missing link" in sustainability implementation [18,19] or voice critical concerns regarding overall resource depletion [20,21] and greenwashing risks in the sustainability domain [22,23].

In spite of large volumes of communication and research regarding sustainability in logistics and supply chain management, empirical evidence for an improvement in global or company-based



www.mdpi.com/journal/sustainability

sustainability is largely missing or simply anecdotal evidence. Quantitative research is, for example, of interest for the productivity of LSP regarding the established triple bottom line approach for sustainability, including economic, environmental, and social performance areas. This is especially true for a dynamic time-series perspective, as usually only static analyses for one point in time are presented. Therefore, for example, the operations research (OR) technique of a data envelopment analysis (DEA) Malmquist index calculation could be used in order to provide a longitudinal calculation of efficiency for multiple objectives regarding the triple bottom line approach for European LSP. The underlying DEA technique for efficiency measurement was introduced by Charnes, Cooper, and Rhodes [24] in the basic form with constant returns to scale (CRS) model and extended by Banker, Charnes, and Cooper [25] with a variable returns to scale (VRS) model. It is well applied within the logistics service sector, where service quality aspects and other competitive features lead to a multi-output production situation, especially in complex supply chain environment or if public interest or public production is included [26–30]. The DEA Malmquist index for the analysis of longitudinal developments [31–33] has also been applied in logistics, e.g., for the airline and airport service sector [34–37].

For the question of longitudinal analysis regarding sustainability developments, some applications and results exist for the energy industry and the aggregated industry as well as on a national country level. For the energy sector, Arabi et al. [38] show that in Iranian energy production, eco-efficiency has increased mainly due to technical advancement and that eco-efficiency and corporate efficiency are closely linked; this is based on a slacks-based model for Malmquist-Luenberger index measurement in the timeframe from 2003 to 2010, a period with huge restructuring efforts. Chang, Kuo, and Chen [39] describe dynamic Malmquist index results regarding three years for the aggregation level of whole industries, reporting that sustainability performances improved overall but that only seven out of 16 observed industries gained in sustainability efficiency throughout the whole three-year period. Lin, Chen, and Chen [40] outline Malmquist-Luenberger index results for 70 countries regarding sustainability performance between 1981 and 2007, revealing that developing countries were able to achieve a higher improvement in their environmental productivity.

However, no application exists to date addressing sustainability questions for LSP with the DEA Malmquist index technique. Therefore this is constituted as a significant research gap. The research questions therefore are how DEA Malmquist index calculation can be applied to the question of longitudinal efficiency development of LSP regarding the sustainability triple bottom line and which results such an analysis would present.

The remainder of this contribution is structured as follows: Section 2 describes management concepts and internationally acknowledged evaluation regimes of sustainability for LSP. Section 3 outlines the policies and data (2007–2015) for selected European LSP regarding sustainability, which were used for evaluation in this contribution. Section 4 provides the methodological background regarding DEA and the DEA Malmquist Index. Section 5 presents calculation results for the dataset. Section 6 points at interesting discussion topics resulting from the calculations results before Section 7 closes with conclusions and further interesting research questions.

2. Transportation and Sustainability

Sustainability has been defined by von Carlowitz regarding the "sustainable" growth and cutting management of forests with the simple rule not to cut more trees in a period than can grow back in a long-term perspective [41]. Similarly, the modern-day scientific as well as practical definition, based on earlier works like "Limits to Growth" [42,43], stems from the Brundtland report in 1987 [44] (p. 16, §27), stipulating that:

"Humanity has the ability to make development sustainable to ensure that it meets the needs of the present without compromising the ability of future generations to meet their own needs. The concept of sustainable development does imply limits—not absolute limits but limitations imposed by the present state of technology and social organization on



environmental resources and by the ability of the biosphere to absorb the effects of human activities. But technology and social organization can be both managed and improved to make way for a new era of economic growth."

As several researchers note, this has been a major step forward, reconciling economic, environmental, and social objectives, a hallmark still valid and approachable today, 30 years later [45–47]. Derived from this overall global and societal objective, corporate strategies and decisions are obviously affected at the core to the point where, as with marketing, quality, and supply chain management in the past, today sustainability can be seen as a fundamental management philosophy permeating the whole corporation and embedded in all processes and decisions [48,49].

Regarding the environmental dimension for transportation, logistics, and supply chain management, the specific challenge consists in the fact that this sector, in which logistics service providers are active, has developed into the largest contributor of greenhouse gas emissions among all economic and society sectors (contributing 25.5% of total EU-28 greenhouse gas emissions in 2014, [50]) and the only one with a dynamic long-term increase in emissions (cp. Figure 1). This is aggravated by the fact that, only hampered by the economic crisis after 2008, the overall development trend for emissions from this sector is still largely upward, endangering the greenhouse gas (GHG) emission reduction targets set by the European Union until 2050 (cp. Figure 2). Emissions from transport (including aviation) in 2014 were 20.1% above 1990 levels; international aviation experienced the largest percentage increase over 1990 levels (+97%), followed by international shipping (+24%) and road transport (+17%, [50]).



Figure 1. Greenhouse Gas (GHG) emission development by sector 1990–2014 [50]. * Transport includes international aviation but excludes international maritime; ** Other includes fugitive emissions from fuels, waste management, and indirect CO_2 emissions.

Furthermore, it can be recognized that there are large differences in such developments as depicted in Figure 3 for the country level development within the European Union. Whereas Nordic states like Lithuania, Finland, Estonia, or Sweden have succeeded in reducing GHG emissions from transportation, other countries have even higher increases than the EU average, led by the Czech Republic, Luxembourg, Ireland, and Poland. Again, changes are only possible on a corporate level.





Figure 2. GHG emission development 1990–2014 and objectives transportation [51].



Figure 3. GHG emission development transport sector 1990–2014 by country [51].

What is the bottom-line for an individual LSP? Analogous to the conclusion drawn by Levallois [52] as well as the European Commission in the White Paper Transportation [53], de-growth or limitation of growth would not be an option as it would counteract the idea of balanced development also in an economic and social dimension. Instead, LSP strive to improve within all three dimensions as per their individual capabilities, e.g., by aiming to reduce the GHG emission volume per tonkilometer of transport service as the overall transport volume and therefore the absolute total GHG emission is out of their individual reach; see, for example, the "Carbon Efficiency Index" by Deutsche Post DHL [8] (p. 76). This is also exemplified by specific triple bottom line missions within corporate strategies and reports (Figure 4). Many research avenues and projects therefore address such questions, mainly aiming to reduce and optimize the logistics "footprint" [54–56]. As was experienced involuntarily during the 2008 economic crisis, the total amount of GHG emissions can easily be reduced by lowering the overall transport volume. This is a political question at the core, not a corporate one, and further complicated by the Jevons Paradox [57] problem as outlined by [58–60]. However, furthering a relative measurement approach, this contribution will try to establish relative measures of sustainability



performance for LSP, e.g., by comparing GHG output volumes to assets and other corporate input volumes. Therefore, in another size-related dimension, the question of corporate size and absolute transport volume does not come into play with such a comparative analysis.



Figure 4. Example corporate sustainability missions DB [61] (pp. 2, 28).

3. Sustainability Policies and Data for European LSP

3.1. Economic Sustainability

Sustainability has been made specific and applicable among other research contributions by the concept of the triple bottom line according to Elkington [62] as well as further extensions and adaptions [63–65]. Therein the established corporate perspective of economic viability and success ("profit" perspective) is widely acknowledged and aim of many research and management science optimization quests [66–68]. One of the main questions still quite unanswered in this context is the question of an adequate timeline horizon for economic planning and evaluation; does economic sustainability therefore mean a profit perspective for one, five, 10, or 50 years or even more [69–72]? It may be recognized as a sideways glance that, with an increasing timeline perspective to be established in this dimension, the areas of economic and environmental sustainability are increasingly merging as, for example, the problem of resource availability and affordability will play a crucial role in any business case for a horizon of, say, 50 years. If there is environmental unsustainability in terms of depletion of resources, the economic sustainability of a corporation connected to such a resource is in vain. For the transportation and logistics sector, in such a perspective, the resource availability of crude oil is of especially high importance as, in spite of many technology advances and political initiatives regarding the use of electric and liquefied natural gas (LNG) or other alternative propulsion systems, the diesel engine will still be a major element of transportation for at least 50 years to come [73–75]. Alas, transportation and logistics companies such as airlines will depend heavily on the availability and price development of crude oil as far as their economic value and



- EBIT—The yearly corporate profit sustainability are concerned [76,77]. Regarding this economic dimension in the proposed efficiency development analysis conducted herein, the following two output types are used as indicators:as measured by the earnings before interest and tax (EBIT) standardized figure in corporate reporting [78,79]. All currency quantities are converted to Euro values according to OECD historical exchange rate data [80].
- Dividend volume—As a further yearly economic performance indicator, the total dividend volume paid to the shareholders is used.

3.2. Environmental Sustainability

The environmental perspective of "planet" within the triple bottom line approach is in the vast majority of research contributions concerning itself with the climate change and carbon emission, respectively greenhouse gas (GHG) emission, problem area as outlined by Wang and He [81] for regional transportation sectors in China, Voytenko et al. [82] for the development of sustainable cities in Europe, or Chavez-Baeza and Sheinbaum-Pardo [83] for the question of sustainable passenger road transport in Mexico City. Regarding environmental analysis for the subsequent efficiency development analysis, the following output type is used as indicator:

Carbon emission volume—The measured and reported scope of two GHG emissions in millions of tonnes of carbon dioxide equivalents are used as a valid output indicator for environmental sustainability [84–86].

3.3. Social Sustainability

A more or less neglected field of the triple bottom line-approach, at least in quantitative research endeavors and within the logistics domain, is the social dimension of "people", see [87,88]. Herein, mainly working rights, safety as well as equal and fair pay questions are addressed [89–92]. As an example for relevant LSP reporting, DHL quotes the number of working accidents within its processes in a static and dynamic year-by-year perspective, e.g., for 2015 as an accident rate of 4.0 accidents per 200,000 h worked with an average of 15.6 working days lost per accident [8] (p. 75). A similar detailed commitment to the social sustainability commitment of employers can be recognized with the example of Panalpina in the following Figure 5.



Figure 5. Example of a corporate sustainability statement [93] (p. 50).



However, in most sustainability discussions, the social dimension is rooted in the corporate role as employer, providing income, participation, and status to people around the world. However, again, this is not the whole story. Moreover, diversity and gender questions are also increasingly addressed in this domain [94–96]. An example is the corporate report section from DHL as follows [8] (p. 74):

"Seeing diversity as potential—People from various cultures, with different points of view and skills work together successfully at Deutsche Post DHL Group. It's precisely this diversity that we see as potential for the further growth of our organization and an opportunity to attract and retain talent. We promote inclusion and reject any form of discrimination—principles are part of your Group-wide Code of Conduct. In 2015, we continued to make a concerted effort to strengthen the significance of diversity within our company. We trained employees, continued the Diversity Council's work and held another global Diversity Week. As at 31 December 2015, the proportion of women in management worldwide was 20.7% (previous year: 19.3%), a figure we intend to raise continuously."

Diversity and gender questions are a quite new topic, as can be seen by the example of the LSP Panalpina with headquarters in Basel, Switzerland: in 2015, the corporate report presents a 19% share of women in senior management, whereas the 150-page 2012 corporate report does not contain the word women once [93,97]. Other companies like Kühne + Nagel are only reporting the share of women in the total employee number (interestingly, constantly 32% throughout 10 years of reporting, [9,98–106]). Regarding social analysis, the following two output types are used as indicators and included in the subsequent analysis:

- Employment—As the core social role for companies is to provide work and income, the full time equivalent (FTE) figure from the corporate LSP reports (usually: figure at year end/31.12.) is taken as the output type for social sustainability.
- Gender equality and women participating in management—Though there are various definitions regarding the respective management level, most LSP comment on the share of women in management positions, be it management in general or "senior management" [92] (p. 49). This share is valued as an important and comparable indicator for the social sustainability and success of LSP in this evaluation. In some cases where the reported numbers are not available for all subsequent year of analysis, the older numbers were assumed to be at least at the level of the latest reported figure. This secures that no companies have disadvantages in this point given the fact that these shares are continuously rising in tendency.

3.4. Input Indicators and Data

Regarding a sensible input volume indicator, the analysis based on corporate reports by the available large European LSP led to the following two feasible indicators:

- Asset volume—The total asset value as reported in the financial reporting of the selected LSP is used. The asset volume for Deutsche Post DHL had to be adjusted due to the former incorporation of the financial institution "Postbank AG" from 2006 to 2008. The (known) asset volume of the banking part was subtracted from the total asset volume in order to reach comparable levels.
- Revenue volume—The total revenue volume is included as indicator for corporate and activity size for input evaluation.

3.5. Company Selection: European Focus

The European region within a global logistics market is singled out for three reasons. First, within the European business environment, a large number of companies are active within a comparable corporate reporting and business environment, enhancing the comparative strength of the analysis. Furthermore, though the industry is built up mainly out of very small and



medium-sized companies, in Europe there are at least some large integrators. Finally, those larger companies are required to publish detailed business and corporate data in their financial reporting according to international standards, which is not the case in other regions like Asia.

For detailed evaluation and analysis, the five largest European forwarders and LSP are selected according to different market share and leadership sources: Deutsche Bahn/Schenker (DB Schenker AG is part of the Deutsche Bahn AG (DB) concern, fully in the ownership of the German federal government; though DB Schenker AG is only contributing 38% to the total DB revenue volume [61] (p. 94), as there are no separate numbers for DB Schenker AG, the comprehensive numbers for Deutsche Bahn AG are applied), Deutsche Post DHL, DSV, Kühne + Nagel, and Panalpina. In some areas like road transportation, Dachser would have to be included, but there are no comprehensive corporate reports publicly available as this is one of the largest family-owned businesses. The Danish logistics giant A.P. Moeller Maersk would have been a further candidate due to its strong position in shipping, but the similarly large business segment of crude oil drilling, transportation, and sale is largely inseparable for most data in the company reporting [10], and therefore, due to comparability reasons, the company was not included. Bringing all input and output types and indicators together, Figure 6 depicts the overall production setup as represented by all selected criteria. Table 1 outlines the major characteristics and correlations of these indicators to be compared to the following DEA Malmquist index calculation results (Section 5); the comprehensive dataset is enclosed in Appendix A.



* Due to input optimization nature (objective: reduction of GHG emission), in the DEA calculation GHG emissions are defined as inputs; further correlation numbers see Table 1.

Figure 6. Input-output diagram for sustainability efficiency evaluation.

Table 1. Input and output indicator characteristics and correlations [8,9,61,97–142] (n = 50).

	Assets (mil. Euro)	Revenue (mil. Euro)	EBIT (mil. Euro)	Dividend Volume (mil. Euro)	GHG Emissions Scope 2 (mil. tonnes CO ₂)	Employment (FTE)	Women in Management (%)
Min	1243.02	3942.79	-1979.00	0.00	38,200.00	13,773.00	10.00
Max	56,059.00	60,545.00	2965.00	1087.00	23,400,000.00	463,350.00	20.70
Arith. Mean	19,124.60	23,200.88	952.48	337.39	6,385,365.49	161,728.16	13.92
Median	4636.40	15,266.49	443.16	198.81	3,260,383.79	68,964.50	13.50
Std.Dev.	20,478.07	19,354.80	1078.47	343.51	8,289,207.12	167,365.53	3.84
Correl. Assets	1.00	0.82	0.79	0.64	0.90	0.83	0.30
Correl. Revenue		1.00	0.72	0.92	0.51	0.99	0.34
Correl. EBIT			1.00	0.63	0.66	0.69	0.28
Correl. Dividend	l			1.00	0.28	0.89	0.29
Correl. GHG					1.00	0.52	0,09
Correl. Employn	nent					1.00	0.36
Correl. Women in	n M.						1.00



4. DEA Malmquist Technique for Longitudinal Efficiency Analysis

The data envelopment analysis (DEA) method is based on the work of Koopmans regarding the activity analysis concept [143], Debreu [144] and Farrell in terms of the radial efficiency measure [145], and the works of Diewert [146]. This led to the specific DEA method suggestion by Charnes, Cooper, and Rhodes [24]. The advantages of its increasing use in higher education research are the fact that no a priori existing knowledge of a production function is required, only real-life data is used (no hypothetical production function or efficiency values), and a multitude of inputs can be combined with a multitude of outputs, which is very typical for LSP, as complex and high-level service providers as well as other industries like health care or higher education [147–151] (Karsak and Karadayi 2017; Bolli et al. 2016; Shwartz, Burgess, and Zhu 2016; Hollingsworth 2003; Bessent et al., 1983). DEA studies Decision Making Units (DMUs), which can be seen as the entities responsible for input, throughput, and output decision making [152] (p. 22); DMUs like LSP can be evaluated and compared, showing a specific level of decision making success in terms of overall efficiency. DEA uses a non-parametric mathematical programming approach for this evaluation of DMUs' efficiency relative to each other (comparing each DMU's relative efficiency with the best DMU). Further it is assumed that there are several DMUs, and it is supposed that inputs and outputs comply with these requirements: (i) For each input and output, there are numerical, positive data to be chosen for all DMUs; (ii) Selected values (inputs, outputs, and the chosen DMUs) should depict the interest of decision-makers towards the relative efficiency evaluations; (iii) DMUs are homogenous in terms of identical inputs and outputs; (iv) Input and output indicator units and scales are congruent. Furthermore, two different basic models can be distinguished; the CCR model, named after the authors Charnes, Cooper, and Rhodes [24], with constant returns to scale and the BCC model with variable returns to scale (named after Banker, Charnes and Cooper) [25]. For both models and their efficiency measurement, the following specifications are made [153] (p. 239):

n	the number of DMUs to be evaluated
---	------------------------------------

DMU_i the j-th DMU

m the number of inputs to each DMU

s the number of outputs to each DMU

x_{ii} amount of the ith input consumed by DMU j

- y_{ki} amount of the kth output produced by DMU j
- eff abbreviation for efficiency
- v_i the weight assigned to the ith input
- u_k the weight assigned to the kth output

$$\text{eff DMU}_{jo} = \frac{\sum_{k=1}^{s} \mu_k \, y_{kjo}}{\sum_{i=1}^{m} v_i \, x_{ijo}} \tag{1}$$

One basic characteristic of the CCR model is the multi-output and multi-input reduction to a single (weighted) input and output combination for each DMU. For a certain DMU, measuring its efficiency and comparison with other DMUs in the system is enabled. Usually executed by a series of linear programming formulations, the single DMU's performance comparison facilitates a ranking of the different analyzed DMUs and scales their relative efficiency from low to high, whereby the last one is defined as 100% efficient. The CCR model contains both mathematical maximization and minimization problems. Detecting DMUs' relative technical efficiency requires on the one hand detection of each DMU's technical efficiency and, on the other hand, the comparison of all DMU's efficiencies. These steps are executed in DEA's simultaneous arithmetic operation. The calculation of the DMU's efficiency value results from the consideration of the weighted inputs and weighted outputs. With the help of quantified inputs and quantified outputs, DEA generates via a quotient one single efficiency ratio for each DMU. The weighting factors are endogenously determined and allow the pooling of heterogeneous inputs and outputs with different units of measurement in one efficiency ratio. Hereby, each DMU's weights are considered optimally in order to maximize the efficiency value



and determine only the definitely provable inefficiency. The analysis for the maximization problem is attached to each DMU, which means that for n DMUs, the quotient needs to be solved. This is done by transforming the quotient into a linear programming problem. With the help of the following figure, the different scales of CCR and BCC are depicted in the case of a single input and a single output.

Based on early works of Malmquist [31], Caves, Christensen, and Diewert [154] proposed a calculation of a productivity index in order to shed light on efficiency changes over time. This new longitudinal perspective is especially promising for the DEA method as most efficiency measurement approaches are directed towards the question of efficiency improvement. In order to provide usable information for this objective, the index is therefore distinguishing between a technological progress for the whole set of DMUs, as for example universities, individually adapted for each institution, on the one side and the technological efficiency on the other side [32]. This second technological efficiency is caused by the organizational and process setup of an institution.

The following formula depicts the mathematical distance function algorithm used for the Malmquist Index [33]:

$$M_{O,CRS}(x_{t}, y_{t}; x_{t+1}, y_{t+1}) = \frac{D_{O,CRS}^{t+1}(x_{t+1}, y_{t+1})}{D_{O,CRS}^{t}(x_{t}, y_{t})} \times \left[\frac{D_{O,CRS}^{t}(x_{t}, y_{t})}{D_{O,CRS}^{t+1}(x_{t}, y_{t})} \times \frac{D_{O,CRS}^{t}(x_{t+1}, y_{t+1})}{D_{O,CRS}^{t+1}(x_{t+1}, y_{t+1})}\right]^{1/2}$$
(2)

A series of applications do exist for this specific longitudinal efficiency analysis, as shown for example by Thanassoulis, Shiraz, and Maniadakis [155] for water and gas companies from different countries, Emrouznejad and Yang [156] for CO_2 emissions of Chinese manufacturing, or Sueyoshi and Goto [157] for the environmental evaluation of petroleum companies. This highlights also the advantage of the Malmquist index in allowing for an efficient development of decomposition, i.e., compared to DEA Window analysis or other efficiency evaluation methods.

5. Research Results

The calculation results have been obtained for the presented input and output data in the timeframe 2006 to 2015 while applying an output-oriented DEA Malmquist index model with the software BANXIA Frontier Analyst for the analysed European LSP.

Due to specific calculation setups and evaluation perspectives, the following 16 different DEA application runs were implemented in this research paper (see Table 2). The variation of constant versus variable returns to scale ("CRS/VRS") refers to the model differences outlined above in Section 4 regarding the production scale assumptions. As there are no similar calculations for LSP known, it makes sense to calculate and compare both model types. The further question of one or two input types relates to the fact that the input type assets may be scrutinized for application with LSP due to the question of different strategy setups ("asset-light" versus "asset-heavy" strategies, as discussed later in Section 6). Finally, the question of weighting is based on the theory regarding the suitable size of the DMU as well as the question of a sensible discrimination level in combination with the concept of the balanced triple bottom line discussed here.

In order to achieve sensible discrimination levels [158] as well as paying attention to the idea of a balanced triple bottom line sustainability evaluation, weightings are introduced as otherwise (and seen in the 0% weightings case), due to the methodological nature of the DEA technique [24,159], nearly all LSP and annual datasets are deemed to be 100% efficient as with the current low number of DMUs compared to the relatively high number of input and output types. Each DMU may find a "specialized" output type at which to excel. Therefore, forcing the DEA technique to acknowledge at least every applicable input and output type with a certain minimum weighting is seen as a sensible variation idea (5%, 10%, and 15% weightings applied, Table 2).



	VRS	(Variable Returns to Scale)	CRS	(Constant Returns to Scale)
Weighting Inputs/Outputs ¹	2 Inputs	1 Input	2 Inputs	1 Input
0% Weighting—Run	(I)	(II)	(III)	(IV)
Arithm. Mean	99.75%	99.75%	99.36%	98.87%
Median	100.00%	100.00%	100.00%	100.00%
Minimum	87.60%	87.60%	74.20%	71.50%
St. Dev.	1.74%	1.74%	3.68%	4.62%
5% Weighting—Run	(V)	(VI)	(VII)	(VIII)
Arithm. Mean	99.49%	99.56%	98.01%	97.76%
Median	100.00%	100.00%	100.00%	100.00%
Minimum	79.10%	79.30%	71.40%	69.10%
St. Dev.	2.97%	2.90%	5.42%	5.89%
10% Weighting—Run	(IX)	(X)	(XI)	(XII)
Arithm. Mean	98.80%	99.09%	85.49%	87.09%
Median	100.00%	100.00%	194.75%	94.00%
Minimum	72.90%	72.80%	43.30%	45.80%
St. Dev.	4.35%	4.10%	17.14%	15.41%
15% Weighting—Run	(XIII)	(XIV)	(XV)	(XVI)
Arithm. Mean	97.57%	98.31%	72.12%	73.93%
Median	100.00%	100.00%	78.60%	76.50%
Minimum	66.60%	67.70%	28.30%	31.00%
St. Dev.	6.30%	5.50%	24.93%	22.18%

Table 2. Data envelopment analysis (DEA) Malmquist index application runs (I-XVI) and characteristics.

Notes: ¹ The applied weightings are not applicable to the output types earnings before interest and tax (EBIT) and dividend volume as in these cases zero values are experienced, which would lead to infeasible zero values for efficiency values in return due to the linear programming nature of the DEA calculation. Cumulated minimum weightings may not exceed 100%, and with 100% itself, the efficiency weights allocation room for the DEA calculation is zero. Therefore the last and maximum weighting applied for the five feasible indicators (assets, revenue, greenhouse gas (GHG) emissions, employment, and women participating in management) is 15%, cumulating to a value of 75% of fixed weighting within all indicators; in other words, the individual efficiency calculation "free float" is only 25%, attributable individually to all seven indicators.

As expected, the results show that, for the aggregate efficiency results (arithmetic mean, median, minimum, and standard deviation values for all 50 calculated data items per run), the differentiation level rises with (i) increasing weightings; (ii) CRS over VRS model applications; and (iii) one input versus two input types. This is explained according to a theory of decreasing "freedom" along the lines of (i), (ii), and (iii) for the individual efficiency optimization and calculation for each DMU in the subsequent models from run I to run XVI. According to the method theory, see [160,161], as VRS models are wielding different results from the CRS models, the supposition of variable returns to scale (VRS) is the more appropriate model application in this specific scenario and dataset. Regarding the number of inputs, as there are only slight differences experienced with this distinction, the application of two inputs seems to be legitimate as this also includes a richer information base (including asset volume). Third, the question of weighting seems to lean towards the maximum possible minimum weighting of 15%, as in these cases (XIII and XIV) the differentiation level is sufficiently high. One might argue that the "fixed" weighting volume of cumulated 75% may be too high and switch to the 10% (cumulative 50%) variant (runs IX and X), but, again, this does not signifies a major difference. Run XIII is exemplified further for discussion, especially regarding the longitudinal perspective.

Quantitative results for all 16 calculation runs are outlined in detail in Appendix B at the end of the paper. Here the main results are provided in an aggregated figure outline as the basis for the subsequent discussion of results (Table 3 and Figure 7). From the results depicted in Table 3, it is obvious that at least in a VRS case, as applicable here, the large number of output types as required by a triple bottom line perspective allows most DMU to reach a 100% (1.00) efficient evaluation over all indicators, even in the here applied minimum weighting of 15% for five indicators applied here (asset and revenue volume, GHG emissions, employees, and women participating in management). Only Deutsche Bahn (Schenker) and DSV are not able to achieve the production frontier in this case



with decreasing (DB Schenker) and increasing returns to scale (DSV) positions respectively. This would translate into the management advice that DB could increase its sustainability efficiency by reducing the overall size and production volume, whereas DSV is too small to reach efficient levels and therefore should increase corporate size and production, meaning transportation service, volume.



Figure 7. DEA Malmquist index developments for run XIII (**top**: Malmquist; **middle**: Catchup; **bottom**: Frontier Shift).



Year	Deutsche Bahn Score	Deutsche Bahn Scale	Deutsche Post DHL Score	Deutsche Post DHL Scale	DSV Score	DSV Scale	Kühne + Nagel Score	Kühne + Nagel Scale	Panalpina Score	Panalpina Scale
2006	1.00	Constant	1.00	Constant	1.00	Constant	1.00	Constant	1.00	Constant
2007	1.00	Constant	1.00	Constant	0.93	Decreasing	1.00	Constant	1.00	Constant
2008	1.00	Constant	1.00	Constant	1.00	Constant	1.00	Constant	1.00	Constant
2009	1.00	Constant	1.00	Constant	0.67	Increasing	1.00	Constant	1.00	Constant
2010	1.00	Constant	1.00	Constant	1.00	Constant	1.00	Constant	1.00	Constant
2011	0.99	Decreasing	1.00	Constant	0.81	Increasing	1.00	Constant	1.00	Constant
2012	1.00	Constant	1.00	Constant	1.00	Constant	1.00	Constant	1.00	Constant
2013	0.86	Decreasing	1.00	Constant	1.00	Constant	1.00	Constant	1.00	Constant
2014	0.88	Decreasing	1.00	Constant	0.87	Increasing	1.00	Constant	1.00	Constant
2015	0.87	Decreasing	1.00	Constant	0.91	Increasing	1.00	Constant	1.00	Constant

Table 3. Efficiency results for run XIII per logistics service provider (LSP).

6. Discussion

For a detailed discussion regarding the results presented above, among others, the following items may be worthwhile to explore:

(I) In a qualitative evaluation of the used indicators and data it may be outlined that though many selection and application decisions in this context, e.g., the use of revenue as input indicator, may be counter-intuitive, the specific production theory context in a sustainability evaluation can be argued to be distinctively different from standard input-output and economic analysis settings. Therefore, the use of assets as well as revenue as a form of corporate size in terms of input is justified. Furthermore, the comparative accuracy in these two cases is very high as IFRS and reporting standards prohibit companies from deviating too much from a common ground of definition, analysis, and reporting. For output measures, the economic indicators of EBIT and dividend volume are to be judged viable and sensible too, as the results show. The environmental indicator of scope 2 GHG emissions has proven to be of interest in terms of efficiency development over time but not so in an inter-company comparative perspective. As scope 2 only reports corporate direct and indirect emissions but not emissions by suppliers and subcontractors, different business structures between LSP influence the scope 2 amount too heavily.

For example, whereas Deutsche Post DHL operates its own aircraft fleet and reports high levels of scope 2 GHG emissions due to that, Panalpina does not employ its own planes but only subcontractors and has therefore significantly lower levels of scope 2 GHG emissions (the same is true for DB Schenker versus Kühne + Nagel and partly DSV). Sensible comparative figures are only to be expected with full disclosure of scope 3 GHG emissions throughout the whole supply chain, which are hard to obtain throughout the multi-level subcontractor structures in logistics. This inherent problem with forwarders as brokers of transportation, not necessarily but also possibly including own transport activities, is captured in the model figure for DSV below.

(II) Plain efficiency results show that among the five selected European LSP, DSV is the least resilient and performant regarding sustainability efficiency in a triple bottom line perspective with the implemented indicators. Already having a 0% minimum weighting restriction, meaning each and every input and output indicator may be individually weighted up to 100% for each DMU, there are single annual data points for DSV (2009, followed by 2011) with an inefficient overall calculation value. This is followed by DB (Schenker) as the next least sustainably efficient LSP in the dataset, with inefficient annual values for 2014 and 2015 from a minimum weighting of 5% for the indicators. On the other side, LSP DHL, Kühne + Nagel, and Panalpina only show inefficient data points in the applied CRS models. Significantly, in all cases, Panalpina is selected as the efficient reference technology, singling it out as the most efficient DMU included in this analysis on all accounts. As DHL is additionally experiencing very low levels of efficiency in the VRS model runs, it can be claimed that especially the two medium-sized, asset-light LSP without government ownership and influence, namely Kühne + Nagel and Panalpina (both also legally seated in Switzerland) are the prominent leaders and benchmarks compared to the other LSP. Further research would have to determine which of the differentiating factors (medium size,



asset-light forwarder strategy, and being devoid of government ownership and influence) may be the most important one for the characteristics of an excellent sustainability efficiency as shown here. As a hypothesis for further discussion and research the following proposition may be formulated:

'It is proposed that medium-sized, asset-light LSP without government ownership and influence may have a competitive advantage in reaching high sustainability efficiency levels in a long-term perspective.'

(III) Longitudinal developments over time (Figure 8) furthermore provide the result that obviously the same distinction as recognized before holds true also for longitudinal development. Whereas DSV and partly DB Schenker are experiencing significant inefficiencies and therefore are predominantly occupied with catch-up developments (middle picture), the other three LSP are actively pushing the efficiency frontier (lower picture), e.g., Kühne + Nagel in the year 2008 due to a very resilient economic posture in spite of the global economic crisis; in 2010 Deutsche Post DHL; and in 2011 Panalpina due to significant EBIT and dividend volume increases. Those two different parts are adding up to the overall Malmquist Index development perspective (upper picture), where it can be seen that sustainability efficiency development seems to be a "shaky" business. Obviously for each LSP and business year, different tendencies occur, and no single LSP is able to sustain efficiency improvements throughout all 10 years (nine change years towards the base year 2006 seen here). This can be connected to the standard business cycle and consolidated into a comprehensive further hypothesis as the proposition of a "triple separation business cycle" model, in which, as in Figure 9 below, the typical growth and crisis phases of economic, environmental, and social indicators move in with the same volatility and sequence ("business cycle") but with a significant shift. For example, compared to the economic business cycle of growth and crisis, as experienced within the analysed timeframe of 2005 to 2016, the social indicator of employment is shifted to the future by one to three years. Meaning that, as experienced in the dataset analysed here, the peak of employment does not occur at the same time as the peak of revenues and profit as economic indicators but is due to employment and dismissal restrictions for employers significantly later. Thus, in the case of the analysed LSP here, the peak of employment was around 2009/2010, later than the economic peak before the 2008/2009 crisis. On average, significant workforce reduction obviously needs a time lag of about one to three years depending on (national) workforce regulations, specialization, and qualification of employees and other factors.



Figure 8. Forwarder position and role (example DSV [162], p. 4).





Figure 9. Triple separation business cycle with economic, environmental, and social indicators.

A similar shift but in the other direction is experienced with environmental factors, e.g., emissions for GHG. Already in the first cooling phases of an economic cycle, transportation volumes are reduced significantly as stock levels are not replenished in full in anticipation of an oncoming crisis. Again, this was experienced in the current dataset as GHG emissions were lowest slightly before the 2008 to 2010 business crisis. Such behavior as depicted in Figure 9 would explain the described "shaky sustainability efficiency" development found in the DEA Malmquist index data. As the "optimal" points regarding the three dimensions are distinctively separate in the typical business cycle, it is hard for any LSP to arrive at a steady and balanced sustainability development.

(IV) Furthermore, the crucial and high-influence role of government ownership within the (German) logistics sector is to be reviewed critically. Two out of five analysed LSP, Deutsche Post DHL and Deutsche Bahn/DB Schenker, are partly (DHL: indirectly via KfW 20.9%, see [8], p. 71) or wholly (DB) owned by the German federal government. This can be seen as unwarranted misuse of a competitive market as many research and political critics interpret it [163,164]. Though privatization of these government monopoly actors has been implemented a long time ago, their influence is still large, especially with the network infrastructure problem with Deutsche Bahn, currently financed by the federal government but on long-distance tracks exclusively used by Deutsche Bahn (monopoly). This has to be reviewed closely and critically, as pointed at by the European Commission as well as the German national competition regulator, in order to avoid undue market bias in transportation [165,166]. Depending on such evaluation and judgement, the inclusion of such government-owned entities in a comparative sustainability analysis as conducted here can be put in doubt. As both companies still act in a very competitive and global market environment, this risk is not seen as striking at the moment though. Nevertheless, the fact that, especially for DB (Schenker), the analysed VRS cases indicated a quite seldom "decreasing returns to scale" situation, this should be further discussed and researched. Such "DRS" situations signify that the DMU in question would be able to increase its efficiency posture by decreasing corporate size and production volume. Therefore researchers, cartel watchers, and politicians alike should be inclined to scrutinize Deutsche Bahn/DB Schenker holdings further if potential downsizing and imposed de-segmentation could be of interest for the public as well as the corporation itself in terms of overall efficiency.



(V) Special emphasis may be put on the innovative role of employment indicators, used in this analysis as the social output performance area. Historically, management science has viewed this indicator predominantly as input factor, but, in light of the social sustainability dimension as well as the major importance of the economic and political arena in our societies, it is more than time to re-evaluate this narrow view. As with third party funding for higher education analysis, factors may play distinctive roles as input as well as output indicators. For third party funding, for example, it is acknowledged in university efficiency analysis that this special area of analysis has input characteristics (financing further input factors like personnel or research expenses) but is at the same time in the majority of applications used as a suitable quantitative indicator for research output and excellence, cp. [148]. This could be applied to the question of employment and FTE indicators in a comprehensive sustainability evaluation in the same fashion. Whereas personnel and especially personnel expenses are possibly used as input measures, it shall be established within the social dimension of the sustainability triple bottom line that FTE quantities are also a reliable measure for the social performance and positive impact of any corporation, as suggested here for logistics.

(VI) Finally, even accepting the limited influence of LSP towards the overall setup of global supply chains (e.g., the trend towards longer transportation distances due to sourcing, production, and retail sites farther apart), on a strategic long-term level there may be a further problem. As described by Jevons, the rebound effect may lead to the consequence that, with increased sustainability efficiency, the increased use of transportation resources may lead to an overall absolute increase in negative sustainability impacts. For transportation in general, this is outlined by Wang and Lu [167], Winebrake et al. [168], or Matos and Silva [169]. In order to identify such rebound effects for the specific triple bottom line objectives and indicators suggested here would have to be tested in further research, but a similar rebound effect may not be rejected out of hand in light of the existing research results.

7. Conclusions

This paper has pointed at the research gap of missing longitudinal DEA Malmquist index measurements for the triple bottom line sustainability performance of European LSP. Several important results and conclusions may be highlighted from this first-time application. The initial research question can be answered positively as the presented results show that the DEA Malmquist index application is feasible for LSP in a sustainability efficiency evaluation regarding triple bottom line objectives, and such an application wields interesting results. Further of interest is the fact that medium-sized LSP like Kühne + Nagel as well as Panalpina were reveaked as efficiency leaders in such a triple bottom line sustainability evaluation. In the proposed longitudinal perspective, it is recognizable that all LSP struggle to keep up sustained positive improvements in comprehensive sustainability efficiency; this led to the suggestion of a triple separation business cycle model in which social development lags behind and environmental development precedes economic development indicators for LSP within the business cycle. The suggested indicators like employment for the social dimension of the triple bottom line are found to be very interesting as, especially in a traditional efficiency and production analysis setting, such indicators would have to be used as input types but are employed here as output types. The discussion connected to this question should be continued regarding the further exploration tof a deeper and richer social dimension of the triple bottom line for LSP, as suggested by Wieland, Handfield, and Durach [170]. The established indicators within this DEA Malmquist index calculation, including asset and revenue volume as input types as well as EBIT, dividend, GHG emissions, employment, and women participating in management as output types, have been proven to signify research and management science value in their result output. Other research project should connect to these results and further the discussion regarding appropriate sustainability indicators on the base of the triple bottom line. Also, further trends and developments in the logistics and supply chain management domain as outlined by Fawcett and Waller or Brockhaus et al., see [171,172], would have to be connected to the efficiency developments of European LSP in a comprehensive sustainability setting.



Furthermore, the following five further research items could stem from this initial research regarding a longitudinal sustainability efficiency evaluation of LSP. First, the proposed intersecting and opposing development of the three triple bottom line perspectives during a typical business cycle would have to be tested and verified further. Second, the important influencing factors of corporate size, asset strategy, and government ownership and influence would have to be scrutinized further in subsequent research endeavors. Especially interesting would be which of these three aspects contributes more to explain LSP inefficiencies in a sustainability evaluation perspective. Third, it could be of interest to analyse the proposed "shaky development" further, i.e. if this holds for LSP seated in other world regions as well as during other business cycles outside the timeframe 2006 to 2015 analysed here. Further confirmatory research with larger numbers of LSP included would be important for advancing the frontier of knowledge and management science application for sustainability management within LSP. Fourth, the question of efficiency should be connected to the field of regulation [173], e.g., with the question of whether enhanced regulation would increase or decrease sustainability efficiency, at least in the triple bottom line concept perspective. Fifth, the method comparison between the DEA Malmquist index and the alternative DEA window analysis should be explored further, see [174].

Finally, as indicated in the introduction, the future evaluation and discussion of sustainability performance by LSP should be considered paramount to sustainability management in general as, in a time of global supply chains, LSP are a central key and cornerstone of any value chain for any given product in worldwide markets. This is uniquely depicted by the below map of an example world-spanning LSP network (in this case Kühne + Nagel, Figure 10). Therefore, it can be deemed a special responsibility towards sustainability and our planet for LSP to cater to the needs of human and global sustainability development as far and wide as technology allows.



Figure 10. Global LSP network—example Kühne + Nagel, [9] (p. 43–44).

Conflicts of Interest: The author declares no conflict of interest.



Appendix A

LSP	Year	Assets (mil. Euro)	Revenue (mil. Euro)	EBIT (mil. Euro)	Dividend Volume (mil. Euro)	GHG Emissions Scope 2 (Tonnes CO ₂ -Equivalent) *	Employment (FTE)	Women in Management (%) *
DB	2006	48,440.00	30,053.00	2143.00	341.00	22,690,050.00	229,200.00	10.00
DB	2007	48,529.00	31,309.00	2370.00	333.00	22,850,000.00	237,078.00	11.00
DB	2008	48,193.00	33,452.00	2483.00	147.00	23,230,000.00	240,242.00	12.00
DB	2009	47,303.00	29,335.00	1685.00	347.00	23,400,000.00	239,382.00	13.00
DB	2010	52,003.00	34,410.00	1866.00	500.00	21,100,000.00	276,310.00	14.00
DB	2011	51,791.00	37,979.00	2309.00	500.00	21,630,000.00	284,319.00	16.40
DB	2012	52,525.00	39,296.00	2708.00	525.00	23,170,000.00	287,508.00	16.50
DB	2013	52,894.00	39,107.00	2236.00	200.00	21,160,000.00	295,653.00	16.90
DB	2014	55,883.00	39,728.00	2109.00	500.00	22,090,000.00	295,763.00	17.50
DB	2015	56,059.00	40,403.00	1759.00	700.00	21,760,000.00	297,202.00	18.20
DHL	2006	32,818.00	60,545.00	2282.00	903.00	7,400,000.00	463,350.00	13.50
DHL	2007	32,510.00	54,043.00	1873.00	1087.00	8,000,000.00	453,626.00	13.50
DHL	2008	31,754.00	54,474.00	-1979.00	725.00	6,700,000.00	451,515.00	13.50
DHL	2009	34,738.00	46,201.00	231.00	725.00	5,600,000.00	424,686.00	15.60
DHL	2010	37,763.00	51,388.00	1835.00	786.00	6,000,000.00	418,946.00	14.60
DHL	2011	38,408.00	52,829.00	2436.00	846.00	5,300,000.00	423,502.00	17.60
DHL	2012	33,857.00	55,512.00	2665.00	846.00	5,400,000.00	428,129.00	18.50
DHL	2013	35,461.00	54,912.00	2865.00	968.00	5,600,000.00	434,974.00	19.60
DHL	2014	36,979.00	56,630.00	2965.00	1030.00	5,670,000.00	443,784.00	19.30
DHL	2015	37,870.00	59,230.00	2411.00	1031.00	6,050,000.00	450,508.00	20.70
DSV	2006	2152.58	4284.80	135.89	6.70	2,464,000.00	21,300.00	10.00
DSV	2007	2189.24	4686.11	212.96	6.71	2,464,000.00	21,280.00	10.00
DSV	2008	3178.54	5015.32	215.70	6.70	2,464,000.00	25,056.00	10.00
DSV	2009	2978.85	4846.33	61.78	0.00	3,212,000.00	19 <i>,</i> 211.00	10.00
DSV	2010	3099.07	5713.78	222.85	6.98	3,221,000.00	19,199.00	10.00
DSV	2011	3044.47	5853.51	207.04	14.06	3,299,767.58	21,679.00	10.00
DSV	2012	3101.12	6110.27	237.41	25.85	3,444,507.36	21,932.00	10.00
DSV	2013	3097.28	6128.85	239.74	31.51	3,454,984.92	22,021.00	10.00
DSV	2014	3181.53	6527.23	246.54	36.28	3,679,562.05	22,874.00	10.00
DSV	2015	3717.00	6819.83	296.42	37.94	3,844,508.39	22,783.00	10.00

Table A1. Calculation Data 2006–2015.



Table A1. Cont.

LSP	Year	Assets (mil. Euro)	Revenue (mil. Euro)	EBIT (mil. Euro)	Dividend Volume (mil. Euro)	GHG Emissions Scope 2 (Tonnes CO ₂ -Equivalent) *	Employment (FTE)	Women in Management (%) *
KÜHNE	2006	3635.44	11,563.49	382.61	81.99	286,394.00	46,290.00	10.00
KÜHNE	2007	3921.82	12,777.27	422.15	107.82	286,394.00	51,075.00	10.00
KÜHNE	2008	3503.29	13,621.53	464.16	327.31	286,394.00	53,823.00	10.00
KÜHNE	2009	3926.25	11,518.68	393.09	180.00	286,394.00	54,680.00	10.00
KÜHNE	2010	4300.53	14,666.40	553.04	197.62	275,511.00	66,045.00	10.00
KÜHNE	2011	4972.27	15,866.58	606.45	265.58	196,000.00	71,884.00	10.00
KÜHNE	2012	5274.90	17,434.29	531.77	386.44	196,000.00	72,399.00	10.00
KÜHNE	2013	5177.59	17,000.58	618.16	340.35	191,884.00	72,036.00	10.00
KÜHNE	2014	5435.22	17,525.56	674.16	576.20	203,571.00	74,497.00	10.00
KÜHNE	2015	5718.60	19,017.95	796.99	786.67	197,871.00	80,056.00	10.00
PANALPINA	2006	1339.77	4916.10	165.88	31.14	57,147.00	14,304.00	18.00
PANALPINA	2007	1387.68	5263.81	182.14	45.08	57,147.00	15,301.00	18.00
PANALPINA	2008	1243.02	5598.96	121.72	49.07	57,147.00	15,630.00	18.00
PANALPINA	2009	1273.90	3942.79	19.85	29.71	38,200.00	13,773.00	18.00
PANALPINA	2010	1439.78	5185.83	10.86	0.00	44,210.00	14,836.00	18.00
PANALPINA	2011	1728.68	5262.95	140.89	38.06	57,147.00	15,051.00	18.00
PANALPINA	2012	1644.05	5558.84	-33.60	39.74	49,682.00	15,224.00	19.00
PANALPINA	2013	1583.17	5489.51	38.99	42.40	52,921.00	16,010.00	19.00
PANALPINA	2014	1587.84	5520.83	96.31	53.75	50,527.00	15,639.00	19.00
PANALPINA	2015	1618.35	5489.82	109.89	77.92	49,353.00	14,843.00	19.00

Notes: * Italic numbers are exponential calculations or estimates.

المنسلة للاستشارات

Appendix B

						(A) Cal	culation I	Run I–IV	with 0%	6 Minim	um Weig	ht Restri	ctions						
		(I)	VRS	2 In.			(II)	VRS	1 In.			(III)	CRS	2 In.		(IV)	CRS	1 In.	
Year	LSP	Malm. Index	Catch.	Front.	Score	Scale	Malm. Index	Catch.	Front.	Score	Scale	Malm. Index	Catch.	Front.	Score	Malm. Index	Catch.	Front.	Score
2006	DB				1.00	Const.				1.00	Const.				1.00				1.00
2007	DB	1.05	1.00	1.05	1.00	Const.	1.05	1.00	1.05	1.00	Const.	1.05	1.00	1.05	1.00	1.04	1.00	1.04	1.00
2009	DB	0.89	1.00	0.89	1.00	Const.	0.89	1.00	0.89	1.00	Const.	0.93	1.00	0.93	1.00	0.93	1.00	0.93	1.00
2008	DB	0.90	1.00	0.90	1.00	Const.	0.90	1.00	0.90	1.00	Const.	0.93	1.00	0.93	1.00	0.93	1.00	0.93	1.00
2011	DB	1.08	1.00	1.08	1.00	Const.	1.08	1.00	1.08	1.00	Const.	1.02	1.00	1.02	1.00	1.05	1.00	1.05	1.00
2010	DB	1.08	1.00	1.08	1.00	Const.	1.08	1.00	1.08	1.00	Const.	1.06	1.00	1.06	1.00	1.02	1.00	1.02	1.00
2012	DB	1.07	1.00	1.07	1.00	Const.	1.07	1.00	1.07	1.00	Const.	1.06	1.00	1.06	1.00	1.06	1.00	1.06	1.00
2013	DB	0.90	1.00	0.90	1.00	Const.	0.90	1.00	0.90	1.00	Const.	0.90	1.00	0.90	1.00	0.90	1.00	0.90	1.00
2014	DB	0.96	1.00	0.96	1.00	Const.	0.96	1.00	0.96	1.00	Const.	0.96	1.00	0.96	1.00	0.96	1.00	0.96	1.00
2015	DB	0.89	1.00	0.89	1.00	Const.	0.89	1.00	0.89	1.00	Const.	0.89	1.00	0.89	1.00	0.89	1.00	0.89	1.00
2006	DHL				1.00	Const.				1.00	Const.				1.00				1.00
2007	DHL	1.02	1.00	1.02	1.00	Const.	1.02	1.00	1.02	1.00	Const.	1.12	1.00	1.12	1.00	1.12	1.00	1.12	1.00
2008	DHL	0.69	1.00	0.69	1.00	Const.	0.69	1.00	0.69	1.00	Const.	0.90	1.00	0.90	1.00	0.90	1.00	0.90	1.00
2009	DHL	1.03	1.00	1.03	1.00	Const.	1.08	1.00	1.08	1.00	Const.	1.02	1.00	1.02	1.00	1.14	1.00	1.14	1.00
2011	DHL	1.58	1.00	1.58	1.00	Const.	1.58	1.00	1.58	1.00	Const.	1.07	1.00	1.07	1.00	1.10	1.00	1.10	1.00
2010	DHL	1.19	1.00	1.19	1.00	Const.	1.19	1.00	1.19	1.00	Const.	1.10	1.00	1.10	1.00	1.07	1.00	1.07	1.00
2012	DHL	1.08	1.00	1.08	1.00	Const.	1.02	1.00	1.02	1.00	Const.	1.04	1.00	1.04	1.00	1.00	1.00	1.00	1.00
2013	DHL	1.06	1.00	1.06	1.00	Const.	1.07	1.00	1.07	1.00	Const.	1.02	1.00	1.02	1.00	1.06	1.00	1.06	1.00
2014	DHL	1.02	1.00	1.02	1.00	Const.	1.02	1.00	1.02	1.00	Const.	1.00	1.00	1.00	1.00	1.00	1.00	1.00	1.00
2015	DHL	0.91	1.00	0.91	1.00	Const.	0.91	1.00	0.91	1.00	Const.	0.88	1.00	0.88	1.00	0.88	1.00	0.88	1.00
2006	DSV				1.00	Const.				1.00	Const.				1.00				1.00
2007	DSV	1.15	1.00	1.15	1.00	Const.	1.14	1.00	1.14	1.00	Const.	1.07	1.00	1.07	1.00	0.93	0.96	0.98	0.96
2008	DSV	1.02	1.00	1.02	1.00	Const.	1.02	1.00	1.02	1.00	Const.	0.90	1.00	0.90	1.00	0.99	1.05	0.94	1.00
2009	DSV	0.60	0.88	0.69	0.88	Incr.	0.61	0.88	0.69	0.88	Incr.	0.72	0.74	0.97	0.74	0.76	0.72	1.06	0.72
2011	DSV	1.45	1.14	1.27	1.00	Const.	1.37	1.14	1.20	1.00	Const.	1.37	1.35	1.02	1.00	0.95	0.83	1.13	0.83
2010	DSV	1.03	1.00	1.03	1.00	Const.	1.03	1.00	1.03	1.00	Const.	0.96	0.94	1.02	1.00	1.26	1.40	0.90	1.00
2012	DSV	0.82	1.00	0.82	1.00	Const.	0.82	1.00	0.82	1.00	Const.	1.01	1.00	1.01	1.00	1.04	1.17	0.89	0.97
2013	DSV	1.04	1.00	1.04	1.00	Const.	1.04	1.00	1.04	1.00	Const.	1.05	1.06	0.99	1.00	1.00	1.03	0.98	1.00
2014	DSV	0.87	1.00	0.87	1.00	Const.	0.87	1.00	0.87	1.00	Const.	0.98	0.99	0.98	1.00	0.96	0.96	1.00	0.96
2015	DSV	1.03	1.00	1.03	1.00	Const.	1.03	1.00	1.03	1.00	Const.	1.06	1.01	1.05	1.00	1.08	1.04	1.03	1.00

Table A2. Calculation Results DEA Malmquist Index 2007–2015 (Base Year 2006).



Table A2. Cont.

						(A) Cal	culation I	Run I–IV	with 0%	Minim	um Weig	ht Restrie	ctions						
		(I)	VRS	2 In.			(II)	VRS	1 In.			(III)	CRS	2 In.		(IV)	CRS	1 In.	
Year	LSP	Malm. Index	Catch.	Front.	Score	Scale	Malm. Index	Catch.	Front.	Score	Scale	Malm. Index	Catch.	Front.	Score	Malm. Index	Catch.	Front.	Score
2006	KÜHNE				1.00	Const.				1.00	Const.				1.00				1.00
2007	KÜHNE	1.15	1.00	1.15	1.00	Const.	1.15	1.00	1.15	1.00	Const.	1.09	1.00	1.09	1.00	1.09	1.00	1.09	1.00
2008	KÜHNE	1.81	1.00	1.81	1.00	Const.	1.74	1.00	1.74	1.00	Const.	1.69	1.00	1.69	1.00	1.60	1.00	1.60	1.00
2009	KÜHNE	0.75	1.00	0.75	1.00	Const.	0.81	1.00	0.81	1.00	Const.	0.75	1.00	0.75	1.00	0.81	1.00	0.81	1.00
2011	KÜHNE	1.04	1.00	1.04	1.00	Const.	1.04	1.00	1.04	1.00	Const.	1.11	1.00	1.11	1.00	1.39	1.00	1.39	1.00
2010	KÜHNE	1.37	1.00	1.37	1.00	Const.	1.44	1.00	1.44	1.00	Const.	1.33	1.00	1.33	1.00	1.11	1.00	1.11	1.00
2012	KÜHNE	1.05	1.00	1.05	1.00	Const.	1.05	1.00	1.05	1.00	Const.	1.03	1.00	1.03	1.00	1.08	1.00	1.08	1.00
2013	KÜHNE	1.03	1.00	1.03	1.00	Const.	1.03	1.00	1.03	1.00	Const.	1.08	1.00	1.08	1.00	1.04	1.00	1.04	1.00
2014	KÜHNE	1.28	1.00	1.28	1.00	Const.	1.28	1.00	1.28	1.00	Const.	1.26	1.00	1.26	1.00	1.26	1.00	1.26	1.00
2015	KÜHNE	1.18	1.00	1.18	1.00	Const.	1.18	1.00	1.18	1.00	Const.	1.18	1.00	1.18	1.00	1.18	1.00	1.18	1.00
2006	PANALP.				1.00	Const.				1.00	Const.				1.00				1.00
2007	PANALP.	1.20	1.00	1.20	1.00	Const.	1.20	1.00	1.20	1.00	Const.	1.16	1.00	1.16	1.00	1.16	1.00	1.16	1.00
2008	PANALP.	1.00	1.00	1.00	1.00	Const.	1.04	1.00	1.04	1.00	Const.	0.89	1.00	0.89	1.00	0.84	1.00	0.84	1.00
2009	PANALP.	1.00	1.00	1.00	1.00	Const.	0.63	1.00	0.63	1.00	Const.	0.89	1.00	0.89	1.00	0.89	1.00	0.89	1.00
2011	PANALP.	1.00	1.00	1.00	1.00	Const.	1.00	1.00	1.00	1.00	Const.	0.67	1.00	0.67	1.00	1.19	1.00	1.19	1.00
2010	PANALP.	4.86	1.00	4.86	1.00	Const.	4.86	1.00	4.86	1.00	Const.	1.19	1.00	1.19	1.00	0.66	1.00	0.66	1.00
2012	PANALP.	1.00	1.00	1.00	1.00	Const.	1.00	1.00	1.00	1.00	Const.	1.08	1.00	1.08	1.00	0.84	1.00	0.84	1.00
2013	PANALP.	1.00	1.00	1.00	1.00	Const.	1.00	1.00	1.00	1.00	Const.	0.84	1.00	0.84	1.00	1.06	1.00	1.06	1.00
2014	PANALP.	1.00	1.00	1.00	1.00	Const.	1.00	1.00	1.00	1.00	Const.	1.15	1.00	1.15	1.00	1.16	1.00	1.16	1.00
2015	PANALP.	1.00	1.00	1.00	1.00	Const.	0.98	1.00	0.98	1.00	Const.	1.06	1.00	1.06	1.00	1.07	1.00	1.07	1.00
						(B) Calc	ulation R	un V–VI	II with 5	% Minin	num Wei	ght Restr	ictions						
		(V)	VRS	2 In.			(VI)	VRS	1 In.			(VII)	CRS	2 In.		(VIII)	CRS	1 In.	
Year	LSP	Malm. Index	Catch.	Front.	Score	Scale	Malm. Index	Catch.	Front.	Score	Scale	Malm. Index	Catch.	Front.	Score	Malm. Index	Catch.	Front.	Score
2006	DB				1.00	Cons.				1.00	Cons.				0.99				1.00
2007	DB	1.05	1.00	1.05	1.00	Cons.	1.05	1.00	1.05	1.00	Cons.	1.01	1.01	1.00	1.00	1.02	1.00	1.02	1.00
2008	DB	0.91	1.00	0.91	1.00	Cons.	0.90	1.00	0.90	1.00	Cons.	0.94	1.00	0.94	1.00	0.93	1.00	0.93	1.00
2009	DB	0.90	1.00	0.90	1.00	Cons.	0.91	1.00	0.91	1.00	Cons.	0.93	1.00	0.93	1.00	0.94	1.00	0.94	1.00
2010	DB	1.09	1.00	1.09	1.00	Cons.	1.09	1.00	1.09	1.00	Cons.	1.02	1.00	1.02	1.00	1.02	1.00	1.02	1.00
2011	DB	1.11	1.00	1.11	1.00	Cons.	1.10	1.00	1.10	1.00	Cons.	1.07	1.00	1.07	1.00	1.06	1.00	1.06	1.00
2012	DB	1.08	1.00	1.08	1.00	Cons.	1.07	1.00	1.07	1.00	Cons.	1.07	1.00	1.07	1.00	1.06	1.00	1.06	1.00
2013	DB	0.91	1.00	0.91	1.00	Cons.	0.91	1.00	0.91	1.00	Cons.	0.91	1.00	0.91	1.00	0.91	1.00	0.91	1.00



Table A2. Cont.

						(B) Calc	ulation R	un V–VI	I with 5	% Minin	num Wei	ght Restr	ictions						
		(V)	VRS	2 In.			(VI)	VRS	1 In.			(VII)	CRS	2 In.		(VIII)	CRS	1 In.	
Year	LSP	Malm. Index	Catch.	Front.	Score	Scale	Malm. Index	Catch.	Front.	Score	Scale	Malm. Index	Catch.	Front.	Score	Malm. Index	Catch.	Front.	Score
2014	DB	0.94	0.96	0.98	0.96	Incr.	0.95	0.99	0.96	0.99	Incr.	0.94	0.96	0.98	0.96	0.95	0.99	0.97	0.99
2015	DB	0.91	1.04	0.88	0.99	Decr.	0.89	1.01	0.89	1.00	Cons.	0.90	1.03	0.88	0.98	0.90	1.01	0.89	1.00
2006	DHL				1.00	Cons.				1.00	Cons.				0.79				0.81
2007	DHL	1.02	1.00	1.02	1.00	Cons.	1.02	1.00	1.02	1.00	Cons.	1.17	1.16	1.01	0.91	1.17	1.17	1.01	0.94
2008	DHL	0.69	1.00	0.69	1.00	Cons.	0.69	1.00	0.69	1.00	Cons.	0.93	1.04	0.89	0.95	0.94	1.05	0.89	0.99
2009	DHL	1.05	1.00	1.05	1.00	Cons.	1.08	1.00	1.08	1.00	Cons.	1.11	0.94	1.18	0.89	1.12	0.93	1.20	0.92
2010	DHL	1.55	1.00	1.55	1.00	Cons.	1.56	1.00	1.56	1.00	Cons.	1.07	1.12	0.95	1.00	1.06	1.09	0.97	1.00
2011	DHL	1.19	1.00	1.19	1.00	Cons.	1.19	1.00	1.19	1.00	Cons.	1.08	1.00	1.08	1.00	1.09	1.00	1.09	1.00
2012	DHL	1.08	1.00	1.08	1.00	Cons.	1.02	1.00	1.02	1.00	Cons.	0.99	1.00	0.99	1.00	0.98	1.00	0.98	1.00
2013	DHL	1.05	1.00	1.05	1.00	Cons.	1.06	1.00	1.06	1.00	Cons.	1.06	1.00	1.06	1.00	1.06	1.00	1.06	1.00
2014	DHL	1.02	1.00	1.02	1.00	Cons.	1.02	1.00	1.02	1.00	Cons.	1.00	1.00	1.00	1.00	1.00	1.00	1.00	1.00
2015	DHL	0.92	1.00	0.92	1.00	Cons.	0.92	1.00	0.92	1.00	Cons.	0.90	1.00	0.90	1.00	0.89	1.00	0.89	1.00
2006	DSV				1.00	Cons.				1.00	Cons.				1.00				1.00
2007	DSV	1.10	1.00	1.10	1.00	Cons.	1.12	1.00	1.12	1.00	Cons.	1.09	1.03	1.06	1.00	0.95	0.95	1.00	0.93
2008	DSV	1.00	1.00	1.00	1.00	Cons.	1.02	1.00	1.02	1.00	Cons.	0.90	1.00	0.90	1.00	1.00	1.08	0.93	1.00
2009	DSV	0.61	0.79	0.77	0.79	Incr.	0.62	0.79	0.78	0.79	Incr.	0.68	0.71	0.96	0.71	0.73	0.69	1.05	0.69
2010	DSV	1.28	1.26	1.01	1.00	Cons.	1.37	1.26	1.08	1.00	Cons.	1.41	1.40	1.00	1.00	1.30	1.45	0.90	1.00
2011	DSV	0.98	1.00	0.98	1.00	Cons.	0.99	1.00	0.99	1.00	Cons.	0.96	0.90	1.06	0.90	0.95	0.81	1.17	0.81
2012	DSV	0.96	1.00	0.96	1.00	Cons.	0.93	1.00	0.93	1.00	Cons.	1.05	1.11	0.95	1.00	1.03	1.16	0.89	0.93
2013	DSV	1.03	1.00	1.03	1.00	Cons.	1.03	1.00	1.03	1.00	Cons.	1.01	1.00	1.01	1.00	1.00	1.05	0.96	0.98
2014	DSV	0.93	1.00	0.93	1.00	Cons.	0.91	1.00	0.91	1.00	Cons.	0.97	0.92	1.06	0.92	0.96	0.93	1.03	0.91
2015	DSV	1.05	1.00	1.05	1.00	Cons.	1.05	1.00	1.05	1.00	Cons.	1.06	1.08	0.98	1.00	1.07	1.10	0.97	1.00
2006	KÜHNE				1.00	Cons.				1.00	Cons.				1.00				1.00
2007	KÜHNE	1.14	1.00	1.14	1.00	Cons.	1.14	1.00	1.14	1.00	Cons.	1.02	1.01	1.01	1.00	1.02	1.01	1.01	1.00
2008	KÜHNE	1.74	1.00	1.74	1.00	Cons.	1.68	1.00	1.68	1.00	Cons.	1.51	1.00	1.51	1.00	1.46	1.00	1.46	1.00
2009	KÜHNE	0.77	1.00	0.77	1.00	Cons.	0.82	1.00	0.82	1.00	Cons.	0.77	1.00	0.77	1.00	0.86	1.00	0.86	1.00
2010	KÜHNE	1.06	1.00	1.06	1.00	Cons.	1.06	1.00	1.06	1.00	Cons.	1.10	1.00	1.10	1.00	1.10	1.00	1.10	1.00
2011	KÜHNE	1.32	1.00	1.32	1.00	Cons.	1.38	1.00	1.38	1.00	Cons.	1.30	1.00	1.30	1.00	1.36	1.00	1.36	1.00
2012	KÜHNE	1.06	1.00	1.06	1.00	Cons.	1.05	1.00	1.05	1.00	Cons.	1.07	1.00	1.07	1.00	1.07	1.00	1.07	1.00
2013	KÜHNE	1.03	1.00	1.03	1.00	Cons.	1.03	1.00	1.03	1.00	Cons.	1.03	1.00	1.03	1.00	1.03	1.00	1.03	1.00
2014	KÜHNE	1.26	1.00	1.26	1.00	Cons.	1.26	1.00	1.26	1.00	Cons.	1.25	1.00	1.25	1.00	1.25	1.00	1.25	1.00
2015	KÜHNE	1.16	1.00	1.16	1.00	Cons.	1.17	1.00	1.17	1.00	Cons.	1.16	1.00	1.16	1.00	1.16	1.00	1.16	1.00



Table A2. Cont.

						(B) Calcu	ulation R	un V–VII	I with 5	% Minin	num Wei	ght Restr	ictions						
		(V)	VRS	2 In.			(VI)	VRS	1 In.			(VII)	CRS	2 In.		(VIII)	CRS	1 In.	
Year	LSP	Malm. Index	Catch.	Front.	Score	Scale	Malm. Index	Catch.	Front.	Score	Scale	Malm. Index	Catch.	Front.	Score	Malm. Index	Catch.	Front.	Score
2006	PANALP.				1.00	Cons.				1.00	Cons.				1.00				1.00
2007	PANALP.	1.18	1.00	1.18	1.00	Cons.	1.19	1.00	1.19	1.00	Cons.	1.15	1.00	1.15	1.00	1.15	1.00	1.15	1.00
2008	PANALP.	1.00	1.00	1.00	1.00	Cons.	1.04	1.00	1.04	1.00	Cons.	0.90	1.00	0.90	1.00	0.85	1.00	0.85	1.00
2009	PANALP.	0.46	1.00	0.46	1.00	Cons.	0.64	1.00	0.64	1.00	Cons.	0.89	1.00	0.89	1.00	0.90	1.00	0.90	1.00
2010	PANALP.	1.00	1.00	1.00	1.00	Cons.	1.00	1.00	1.00	1.00	Cons.	0.68	1.00	0.68	1.00	0.66	1.00	0.66	1.00
2011	PANALP.	3.19	1.00	3.19	1.00	Cons.	3.66	1.00	3.66	1.00	Cons.	1.17	1.00	1.17	1.00	1.18	1.00	1.18	1.00
2012	PANALP.	1.00	1.00	1.00	1.00	Cons.	1.00	1.00	1.00	1.00	Cons.	0.85	1.00	0.85	1.00	0.85	1.00	0.85	1.00
2013	PANALP.	1.00	1.00	1.00	1.00	Cons.	1.00	1.00	1.00	1.00	Cons.	1.08	1.00	1.08	1.00	1.06	1.00	1.06	1.00
2014	PANALP.	1.00	1.00	1.00	1.00	Cons.	1.00	1.00	1.00	1.00	Cons.	1.14	1.00	1.14	1.00	1.14	1.00	1.14	1.00
2015	PANALP.	1.00	1.00	1.00	1.00	Cons.	0.98	1.00	0.98	1.00	Cons.	1.05	1.00	1.05	1.00	1.07	1.00	1.07	1.00

(C) Calculation Run IX-XII with 10% Minimum Weight Restrictions

		IX	VRS	2 In.			x	VRS	1 In.			XI	CRS	2 In.		XII	CRS	1 In.	
Year	LSP	Malm. Index	Catch.	Front.	Score	Scale	Malm. Index	Catch.	Front.	Score	Scale	Malm. Index	Catch.	Front.	Score	Malm. Index	Catch.	Front.	Score
2006	DB				1.00	Cons.				1.00	Cons.				0.57				0.62
2007	DB	1.06	1.00	1.06	1.00	Cons.	1.06	1.00	1.06	1.00	Cons.	1.01	1.09	0.92	0.62	1.00	1.09	0.92	0.68
2008	DB	0.92	1.00	0.92	1.00	Cons.	0.92	1.00	0.92	1.00	Cons.	0.98	1.13	0.87	0.70	0.97	1.13	0.86	0.77
2009	DB	0.91	1.00	0.91	1.00	Cons.	0.92	1.00	0.92	1.00	Cons.	0.93	1.03	0.90	0.72	0.94	1.02	0.92	0.79
2010	DB	1.09	1.00	1.09	1.00	Cons.	1.09	1.00	1.09	1.00	Cons.	0.99	1.27	0.78	0.91	0.99	1.27	0.78	1.00
2012	DB	1.08	1.00	1.08	1.00	Cons.	1.11	1.00	1.11	1.00	Cons.	1.02	0.80	1.28	0.73	1.01	0.79	1.27	0.79
2011	DB	1.13	1.00	1.13	1.00	Cons.	1.07	1.00	1.07	1.00	Cons.	1.08	1.33	0.81	0.97	1.06	1.26	0.84	1.00
2013	DB	0.91	0.94	0.96	0.94	Decr.	0.92	1.00	0.92	1.00	Cons.	0.87	0.91	0.96	0.88	0.89	0.96	0.92	0.96
2014	DB	0.95	0.96	0.99	0.90	Decr.	0.94	0.94	1.00	0.94	Decr.	0.94	0.87	1.09	0.76	0.94	0.87	1.09	0.83
2015	DB	0.94	1.02	0.92	0.92	Decr.	0.92	1.04	0.89	0.98	Decr.	0.94	0.99	0.95	0.75	0.94	0.98	0.95	0.82
2006	DHL				1.00	Cons.				1.00	Cons.				0.43				0.46
2007	DHL	1.01	1.00	1.01	1.00	Cons.	1.01	1.00	1.01	1.00	Cons.	1.15	1.17	0.98	0.51	1.16	1.18	0.99	0.54
2008	DHL	0.70	1.00	0.70	1.00	Cons.	0.70	1.00	0.70	1.00	Cons.	0.77	1.04	0.73	0.53	0.78	1.05	0.75	0.57
2009	DHL	1.06	1.00	1.06	1.00	Cons.	1.08	1.00	1.08	1.00	Cons.	1.11	0.95	1.16	0.50	1.13	0.94	1.20	0.54
2010	DHL	1.51	1.00	1.51	1.00	Cons.	1.54	1.00	1.54	1.00	Cons.	1.34	1.34	1.00	0.67	1.33	1.35	0.99	0.72
2012	DHL	1.07	1.00	1.07	1.00	Cons.	1.19	1.00	1.19	1.00	Cons.	1.11	0.94	1.18	0.64	1.11	0.94	1.19	0.68
2011	DHL	1.18	1.00	1.18	1.00	Cons.	1.02	1.00	1.02	1.00	Cons.	1.02	1.27	0.80	0.81	1.00	1.27	0.79	0.86
2013	DHL	1.05	1.00	1.05	1.00	Cons.	1.06	1.00	1.06	1.00	Cons.	1.07	0.95	1.13	0.77	1.07	0.95	1.13	0.82



Table A2. Cont.

						(C) Calc	ulation Ru	ın IX–XI	I with 10	0% Mini	num We	ight Resti	rictions						
		IX	VRS	2 In.			X	VRS	1 In.			XI	CRS	2 In.		XII	CRS	1 In.	
Year	LSP	Malm. Index	Catch.	Front.	Score	Scale	Malm. Index	Catch.	Front.	Score	Scale	Malm. Index	Catch.	Front.	Score	Malm. Index	Catch.	Front.	Score
2014	DHL	1.02	1.00	1.02	1.00	Cons.	1.02	1.00	1.02	1.00	Cons.	1.00	0.85	1.18	0.65	1.00	0.85	1.18	0.69
2015	DHL	0.93	1.00	0.93	1.00	Cons.	0.93	1.00	0.93	1.00	Cons.	0.92	1.00	0.92	0.65	0.92	1.00	0.92	0.69
2006	DSV				1.00	Cons.				1.00	Cons.				1.00				1.00
2007	DSV	1.04	1.00	1.04	1.00	Cons.	1.06	1.00	1.06	1.00	Cons.	1.13	1.10	1.03	1.00	0.97	0.97	1.00	0.90
2008	DSV	0.99	1.00	0.99	1.00	Cons.	1.02	1.00	1.02	1.00	Cons.	0.91	1.00	0.90	1.00	1.01	1.10	0.92	0.99
2009	DSV	0.58	0.73	0.79	0.73	Incr.	0.62	0.73	0.86	0.73	Incr.	0.65	0.69	0.95	0.69	0.70	0.67	1.04	0.67
2010	DSV	1.45	1.37	1.05	1.00	Cons.	1.33	1.37	0.97	1.00	Cons.	1.44	1.46	0.99	1.00	1.33	1.48	0.90	0.99
2012	DSV	1.01	1.11	0.92	1.00	Cons.	0.97	0.90	1.08	0.90	Incr.	0.96	0.84	1.14	0.84	0.95	0.79	1.20	0.78
2011	DSV	0.97	0.90	1.07	0.90	Incr.	0.98	1.11	0.88	1.00	Cons.	1.05	1.19	0.88	1.00	1.03	1.15	0.90	0.89
2013	DSV	1.02	1.00	1.02	1.00	Cons.	1.02	1.00	1.02	1.00	Cons.	1.00	0.93	1.08	0.93	1.00	1.03	0.97	0.92
2014	DSV	0.95	1.00	0.95	1.00	Cons.	0.93	1.00	0.93	1.00	Cons.	0.97	0.92	1.06	0.85	0.96	0.91	1.05	0.84
2015	DSV	1.06	1.00	1.06	1.00	Cons.	1.06	1.00	1.06	1.00	Cons.	1.06	1.15	0.92	0.98	1.07	1.19	0.90	1.00
2006	KÜHNE				1.00	Cons.				1.00	Cons.				0.87				0.88
2007	KÜHNE	1.13	1.00	1.13	1.00	Cons.	1.13	1.00	1.13	1.00	Cons.	1.02	0.99	1.02	0.86	1.02	1.00	1.02	0.86
2008	KÜHNE	1.67	1.00	1.67	1.00	Cons.	1.62	1.00	1.62	1.00	Cons.	1.55	1.17	1.33	1.00	1.51	1.16	1.30	1.00
2009	KÜHNE	0.79	1.00	0.79	1.00	Cons.	0.83	1.00	0.83	1.00	Cons.	0.78	1.00	0.78	1.00	0.87	1.00	0.87	1.00
2010	KÜHNE	1.08	1.00	1.08	1.00	Cons.	1.07	1.00	1.07	1.00	Cons.	1.07	1.00	1.07	1.00	1.03	1.00	1.03	1.00
2012	KÜHNE	1.06	1.00	1.06	1.00	Cons.	1.34	1.00	1.34	1.00	Cons.	1.31	1.00	1.31	1.00	1.38	1.00	1.38	1.00
2011	KÜHNE	1.27	1.00	1.27	1.00	Cons.	1.05	1.00	1.05	1.00	Cons.	1.05	1.00	1.05	1.00	1.03	1.00	1.03	1.00
2013	KÜHNE	1.02	1.00	1.02	1.00	Cons.	1.02	1.00	1.02	1.00	Cons.	1.03	1.00	1.03	1.00	1.03	1.00	1.03	1.00
2014	KÜHNE	1.24	1.00	1.24	1.00	Cons.	1.24	1.00	1.24	1.00	Cons.	1.29	1.00	1.29	1.00	1.29	1.00	1.29	1.00
2015	KÜHNE	1.15	1.00	1.15	1.00	Cons.	1.15	1.00	1.15	1.00	Cons.	1.22	0.99	1.24	0.99	1.22	0.99	1.23	0.99
2006	PANALP.				1.00	Cons.				1.00	Cons.				1.00				1.00
2007	PANALP.	1.17	1.00	1.17	1.00	Cons.	1.17	1.00	1.17	1.00	Cons.	1.13	1.00	1.13	1.00	1.13	1.00	1.13	1.00
2008	PANALP.	1.00	1.00	1.00	1.00	Cons.	1.03	1.00	1.03	1.00	Cons.	0.90	1.00	0.90	1.00	0.86	1.00	0.86	1.00
2009	PANALP.	0.53	1.00	0.53	1.00	Cons.	0.66	1.00	0.66	1.00	Cons.	0.89	1.00	0.89	1.00	0.91	1.00	0.91	1.00
2010	PANALP.	1.00	1.00	1.00	1.00	Cons.	1.00	1.00	1.00	1.00	Cons.	0.69	1.00	0.69	1.00	0.67	1.00	0.67	1.00
2012	PANALP.	1.00	1.00	1.00	1.00	Cons.	3.13	1.00	3.13	1.00	Cons.	1.16	1.00	1.16	1.00	1.17	1.00	1.17	1.00
2011	PANALP.	2.62	1.00	2.62	1.00	Cons.	1.00	1.00	1.00	1.00	Cons.	0.86	1.00	0.86	1.00	0.86	1.00	0.86	1.00
2013	PANALP.	1.00	1.00	1.00	1.00	Cons.	1.00	1.00	1.00	1.00	Cons.	1.08	1.00	1.08	1.00	1.06	1.00	1.06	1.00
2014	PANALP.	1.00	1.00	1.00	1.00	Cons.	1.00	1.00	1.00	1.00	Cons.	1.13	1.00	1.13	1.00	1.13	1.00	1.13	1.00
2015	PANALP.	1.00	1.00	1.00	1.00	Cons.	0.98	1.00	0.98	1.00	Cons.	1.05	1.00	1.05	1.00	1.06	1.00	1.06	1.00



Table A2. Cont.

		(D) Calculation Run V–VIII with 15% Minimum Weight Restrictions																	
		(XIII)	VRS	2 In.			(XIV)	VRS	1 In.			(XV)	CRS	2 In.		(XVI)	CRS	1 In.	
Year	LSP	Malm. Index	Catch.	Front.	Score	Scale	Malm. Index	Catch.	Front.	Score	Scale	Malm. Index	Catch.	Front.	Score	Malm. Index	Catch.	Front.	Score
2006	DB				1.00	Cons.				1.00	Cons.				0.37				0.43
2007	DB	1.07	1.00	1.07	1.00	Cons.	1.06	1.00	1.06	1.00	Cons.	1.02	1.11	0.92	0.41	1.01	1.11	0.92	0.48
2008	DB	0.94	1.00	0.94	1.00	Cons.	0.93	1.00	0.93	1.00	Cons.	0.96	1.11	0.86	0.45	0.95	1.11	0.85	0.53
2009	DB	0.91	1.00	0.91	1.00	Cons.	0.92	1.00	0.92	1.00	Cons.	0.98	0.97	1.01	0.44	1.00	0.96	1.04	0.51
2010	DB	1.10	1.00	1.10	1.00	Cons.	1.09	1.00	1.09	1.00	Cons.	1.00	1.25	0.80	0.55	0.99	1.25	0.79	0.64
2011	DB	1.14	0.99	1.16	0.99	Decr.	1.12	1.00	1.12	1.00	Cons.	1.03	0.89	1.16	0.49	1.02	0.88	1.15	0.56
2012	DB	1.08	1.01	1.06	1.00	Cons.	1.07	1.00	1.07	1.00	Cons.	1.07	1.18	0.90	0.58	1.06	1.18	0.90	0.67
2013	DB	0.90	0.86	1.05	0.86	Decr.	0.92	0.94	0.98	0.94	Decr.	0.87	0.95	0.92	0.55	0.88	0.95	0.92	0.56
2014	DB	1.01	1.02	0.99	0.88	Decr.	0.95	0.95	0.99	0.90	Decr.	1.01	0.92	1.09	0.50	1.01	0.92	1.10	0.58
2015	DB	0.96	0.99	0.97	0.87	Decr.	0.95	1.03	0.92	0.93	Decr.	0.97	1.00	0.97	0.50	0.97	1.00	0.97	0.58
2006	DHL				1.00	Cons.				1.00	Cons.				0.28				0.31
2007	DHL	1.01	1.00	1.01	1.00	Cons.	1.01	1.00	1.01	1.00	Cons.	1.13	1.17	0.97	0.33	1.15	1.18	0.97	0.37
2008	DHL	0.71	1.00	0.71	1.00	Cons.	0.71	1.00	0.71	1.00	Cons.	0.66	1.04	0.64	0.35	0.67	1.05	0.63	0.39
2009	DHL	1.08	1.00	1.08	1.00	Cons.	1.09	1.00	1.09	1.00	Cons.	1.10	0.96	1.15	0.33	1.15	0.95	1.21	0.37
2010	DHL	1.48	1.00	1.48	1.00	Cons.	1.51	1.00	1.51	1.00	Cons.	1.53	1.23	1.24	0.41	1.52	1.25	1.22	0.46
2011	DHL	1.17	1.00	1.17	1.00	Cons.	1.18	1.00	1.18	1.00	Cons.	1.13	1.04	1.08	0.43	1.13	1.03	1.09	0.47
2012	DHL	1.07	1.00	1.07	1.00	Cons.	1.02	1.00	1.02	1.00	Cons.	1.03	1.16	0.88	0.49	1.00	1.15	0.87	0.54
2013	DHL	1.05	1.00	1.05	1.00	Cons.	1.05	1.00	1.05	1.00	Cons.	1.06	0.98	1.07	0.49	1.07	0.99	1.08	0.54
2014	DHL	1.01	1.00	1.01	1.00	Cons.	1.01	1.00	1.01	1.00	Cons.	1.00	0.89	1.12	0.43	1.00	0.89	1.12	0.48
2015	DHL	0.94	1.00	0.94	1.00	Cons.	0.94	1.00	0.94	1.00	Cons.	0.92	1.01	0.91	0.44	0.92	1.01	0.91	0.48
2006	DSV				1.00	Cons.				1.00	Cons.				1.00				1.00
2007	DSV	0.94	0.93	1.01	0.93	Decr.	1.01	1.00	1.01	1.00	Cons.	1.12	1.09	1.03	0.93	1.00	0.99	1.01	0.88
2008	DSV	0.92	1.08	0.86	1.00	Cons.	1.02	1.00	1.02	1.00	Cons.	0.94	1.08	0.87	1.00	1.01	1.11	0.92	0.97
2009	DSV	0.58	0.67	0.87	0.67	Incr.	0.58	0.68	0.86	0.68	Incr.	0.62	0.66	0.95	0.66	0.68	0.66	1.02	0.64
2010	DSV	1.53	1.50	1.02	1.00	Cons.	1.44	1.48	0.97	1.00	Cons.	1.44	1.45	1.00	0.95	1.33	1.44	0.92	0.93
2011	DSV	0.96	0.81	1.19	0.81	Incr.	0.96	0.82	1.18	0.82	Incr.	0.96	0.83	1.16	0.78	0.96	0.81	1.18	0.76
2012	DSV	1.03	1.24	0.83	1.00	Cons.	1.01	1.22	0.82	1.00	Cons.	1.04	1.16	0.90	0.91	1.03	1.14	0.90	0.86
2013	DSV	1.01	1.00	1.01	1.00	Cons.	1.01	1.00	1.01	1.00	Cons.	1.00	0.94	1.07	0.85	1.00	0.98	1.02	0.84
2014	DSV	0.95	0.87	1.10	0.87	Incr.	0.94	0.91	1.03	0.91	Incr.	0.97	0.93	1.05	0.79	0.96	0.92	1.04	0.78
2015	DSV	1.06	1.05	1.01	0.91	Incr.	1.07	1.08	0.99	0.98	Incr.	1.05	1.12	0.94	0.88	1.06	1.16	0.92	0.90

Table A2. Cont.

	(D) Calculation Run V–VIII with 15% Minimum Weight Restrictions																		
		(XIII)	VRS	2 In.			(XIV)	VRS	1 In.			(XV)	CRS	2 In.		(XVI)	CRS	1 In.	
Year	LSP	Malm. Index	Catch.	Front.	Score	Scale	Malm. Index	Catch.	Front.	Score	Scale	Malm. Index	Catch.	Front.	Score	Malm. Index	Catch.	Front.	Score
2006	KÜHNE				1.00	Cons.				1.00	Cons.				0.72				0.74
2007	KÜHNE	1.12	1.00	1.12	1.00	Cons.	1.12	1.00	1.12	1.00	Cons.	1.02	0.99	1.04	0.71	1.02	0.99	1.03	0.72
2008	KÜHNE	1.60	1.00	1.60	1.00	Cons.	1.57	1.00	1.57	1.00	Cons.	1.60	1.25	1.28	0.89	1.55	1.25	1.24	0.91
2009	KÜHNE	0.81	1.00	0.81	1.00	Cons.	0.84	1.00	0.84	1.00	Cons.	0.83	1.01	0.82	1.00	0.93	1.00	0.93	0.91
2010	KÜHNE	1.10	1.00	1.10	1.00	Cons.	1.09	1.00	1.09	1.00	Cons.	1.03	1.11	0.93	1.00	0.98	1.10	0.89	1.00
2011	KÜHNE	1.23	1.00	1.23	1.00	Cons.	1.29	1.00	1.29	1.00	Cons.	1.20	0.84	1.43	0.84	1.23	0.81	1.52	0.81
2012	KÜHNE	1.06	1.00	1.06	1.00	Cons.	1.05	1.00	1.05	1.00	Cons.	1.03	1.16	0.89	0.97	1.01	1.20	0.84	0.98
2013	KÜHNE	1.02	1.00	1.02	1.00	Cons.	1.02	1.00	1.02	1.00	Cons.	1.09	0.89	1.23	0.86	1.09	0.89	1.22	0.87
2014	KÜHNE	1.21	1.00	1.21	1.00	Cons.	1.21	1.00	1.21	1.00	Cons.	1.20	0.91	1.31	0.79	1.20	0.92	1.31	0.80
2015	KÜHNE	1.13	1.00	1.13	1.00	Cons.	1.14	1.00	1.14	1.00	Cons.	1.17	0.93	1.25	0.74	1.16	0.93	1.25	0.74
2006	PANALP.				1.00	Cons.				1.00	Cons.				1.00				1.00
2007	PANALP.	1.15	1.00	1.15	1.00	Cons.	1.15	1.00	1.15	1.00	Cons.	1.12	1.00	1.12	1.00	1.12	1.00	1.12	1.00
2008	PANALP.	1.00	1.00	1.00	1.00	Cons.	1.03	1.00	1.03	1.00	Cons.	0.91	1.00	0.91	1.00	0.87	1.00	0.87	1.00
2009	PANALP.	0.59	1.00	0.59	1.00	Cons.	0.68	1.00	0.68	1.00	Cons.	0.89	1.00	0.89	1.00	0.92	1.00	0.92	1.00
2010	PANALP.	1.00	1.00	1.00	1.00	Cons.	1.00	1.00	1.00	1.00	Cons.	0.70	1.00	0.70	1.00	0.68	1.00	0.68	1.00
2011	PANALP.	2.32	1.00	2.32	1.00	Cons.	2.82	1.00	2.82	1.00	Cons.	1.14	1.00	1.14	1.00	1.16	1.00	1.16	1.00
2012	PANALP.	1.00	1.00	1.00	1.00	Cons.	1.00	1.00	1.00	1.00	Cons.	0.87	1.00	0.87	1.00	0.87	1.00	0.87	1.00
2013	PANALP.	1.00	1.00	1.00	1.00	Cons.	1.00	1.00	1.00	1.00	Cons.	1.07	1.00	1.07	1.00	1.05	1.00	1.05	1.00
2014	PANALP.	0.99	1.00	0.99	1.00	Cons.	0.99	1.00	0.99	1.00	Cons.	1.12	1.00	1.12	1.00	1.12	1.00	1.12	1.00
2015	PANALP.	1.00	1.00	1.00	1.00	Cons.	0.98	1.00	0.98	1.00	Cons.	1.04	1.00	1.04	1.00	1.05	1.00	1.05	1.00



References

- 1. Abbasi, M.; Nilsson, F. Developing environmentally sustainable logistics Exploring themes and challenges from a logistics service providers' perspective. *Transp. Res. Part D* **2016**, *46*, 273–283. [CrossRef]
- 2. Govindan, K.; Soleimani, H. A review of reverse logistics and closed-loop supply chains: A Journal of Cleaner Production focus. *J. Clean. Prod.* **2017**, *142*, 371–384. [CrossRef]
- 3. Colicchia, C.; Marchet, G.; Melacini, M.; Perotti, S. Building environmental sustainability: Empirical evidence from Logistics Service Providers. *J. Clean. Prod.* **2013**, *59*, 197–209. [CrossRef]
- 4. Kamga, C.; Yazici, A. Achieving environmental sustainability beyond technological improvements: Potential role of high-speed rail in the United States of America. *Transp. Res. Part D* 2014, *31*, 148–164. [CrossRef]
- 5. Davarzani, H.; Fahimnia, B.; Bell, M.; Sarkis, J. Greening ports and maritime logistics: A review. *Transp. Res. Part D* **2016**, *48*, 473–487. [CrossRef]
- 6. Brix-Asala, C.; Hahn, R.; Seuring, S. Reverse logistics and informal valorisation at the Base of the Pyramid: A case study on sustainability synergies and trade-offs. *Eur. Manag. J.* **2016**, *34*, 414–423. [CrossRef]
- Lee, C.K.M.; Lee Lam, J.S. Managing reverse logistics to enhance sustainability of industrial marketing. *Ind. Mark. Manag.* 2012, 41, 589–598. [CrossRef]
- 8. Deutsche Post DHL. Smart Logistics—2015 Annual Report; Deutsche Post AG: Bonn, Germany, 2016.
- 9. Kühne + Nagel. Annual Report 2015—Tradition and Innovation; Kühne + Nagel: Schindellegi, Switzerland, 2016.
- 10. A.P. Moeller Maersk A/S. Annual Report 2015; A.P. Moeller Maersk A/S: Copenhagen, Denmark, 2016.
- 11. Higgins, C.; Coffey, B. Improving how sustainability reports drive change: A critical discourse analysis. *J. Clean. Prod.* **2016**, 136, 18–29. [CrossRef]
- 12. Kuzey, C.; Uyar, C. Determinants of sustainability reporting and its impact on firm value: Evidence from the emerging market of Turkey. *J. Clean. Prod.* **2017**, *143*, 27–39. [CrossRef]
- 13. Eroglu, C.; Kurt, A.C.; Elwakil, O.S. Stock market reaction to quality, safety, and sustainability awards in logistics. *J. Bus. Logist.* **2016**, *37*, 329–345. [CrossRef]
- 14. Busse, C.; Kach, A.P.; Bode, C. Sustainability and the False Sense of Legitimacy: How Institutional Distance Augments Risk in Global Supply Chains. *J. Bus. Logist.* **2016**, *37*, 312–328. [CrossRef]
- 15. Tate, W.L.; Dooley, K.J.; Ellram, L.M. Transaction cost and institutional drivers of supplier adoption of environmental practices. *J. Bus. Logist.* **2011**, *32*, 6–16. [CrossRef]
- 16. Dijkema, G.P.J.; Ferra, P.; Herder, P.M.; Heitor, M. Trends and opportunities framing innovation for sustainability in the learning society. *Technol. Forecast. Soc. Chang.* **2006**, *73*, 215–227. [CrossRef]
- 17. Heiskala, M.; Jokinen, J.-P.; Tinnilä, M. Crowdsensing-based transportation services—An analysis from business model and sustainability viewpoints. *Res. Transp. Bus. Manag.* **2016**, *18*, 38–48. [CrossRef]
- 18. Grandia, J. Finding the missing link: Examining the mediating role of sustainable public procurement behavior. *J. Clean. Prod.* **2016**, *124*, 183–190. [CrossRef]
- 19. Al-Rawahy, K.H. Engineering Education and Sustainable Development: The Missing Link. *Procedia Soc. Behav. Sci.* **2013**, *102*, 392–401. [CrossRef]
- 20. Kuhn, M.A.; Kuhn, P.; Villeval, M.C. Decision-Environment Effects on Intertemporal Financial Choices: How Relevant are Resource-Depletion Models? *J. Econ. Behav. Organ.* **2017**, *137*, 72–89. [CrossRef]
- 21. Rimos, S.; Hoadley, A.F.A.; Brennan, D.J. Environmental consequence analysis for resource depletion. *Process Saf. Environ. Prot.* **2014**, *9*, 849–861. [CrossRef]
- 22. Siano, A.; Vollero, A.; Conte, F.; Amabile, S. 'More than words': Expanding the taxonomy of greenwashing after the Volkswagen scandal. *J. Bus. Res.* **2017**, *71*, 27–37. [CrossRef]
- 23. Mahoney, L.S.; Thorne, L.; Cecil, L.; LaGore, W. A research note on standalone corporate social responsibility reports: Signaling or greenwashing? *Crit. Perspect. Account.* **2013**, *24*, 350–359. [CrossRef]
- 24. Charnes, A.; Cooper, W.W.; Rhodes, E.L. Measuring the efficiency of decision making units. *Eur. J. Oper. Res.* **1978**, *2*, 429–444. [CrossRef]
- 25. Banker, R.D.; Charnes, A.; Cooper, W.W. Some models for estimating technical and scale inefficiencies in data envelopment analysis. *Manag. Sci.* **1984**, *30*, 1078–1092. [CrossRef]
- 26. Marchetti, D.; Wanke, P. Brazil's rail freight transport: Efficiency analysis using two-stage DEA and cluster-driven public policies. *Socio-Econ. Plan. Sci.* **2016**. [CrossRef]



- 27. Olfat, L.; Amiri, M.; Soufi, J.B.; Pishdar, M. A dynamic network efficiency measurement of airports performance considering sustainable development concept: A fuzzy dynamic network-DEA approach. *J. Air Transp. Manag.* **2016**, *57*, 272–290. [CrossRef]
- 28. Wanke, P.; Barros, C.P.; Nwaogbe, O.R. Assessing productive efficiency in Nigerian airports using Fuzzy-DEA. *Transp. Policy* **2016**, *49*, 9–19. [CrossRef]
- 29. Zhao, Y.; Triantis, K.; Murray-Tuite, P.; Edara, P. Performance measurement of a transportation network with a downtown space reservation system: A network-DEA approach. *Transp. Res. Part E* **2011**, *47*, 1140–1159. [CrossRef]
- 30. Min, H.; Joo, S.J. Benchmarking the operational efficiency of third party logistics providers using data envelopment analysis. *Supply Chain Manag. Int. J.* 2006, *11*, 259–265. [CrossRef]
- 31. Malmquist, S. Index numbers and indifference surfaces. *Trab. Estad. Investig. Oper.* **1953**, 42, 209–242. [CrossRef]
- 32. Grifell-Tatje, E.; Lovell, C.A.K.; Pastor, J.T. A quasi-Malmquist productivity index. *J. Product. Anal.* **1998**, *10*, 7–20. [CrossRef]
- 33. Wang, Y.-M.; Lan, Y.-X. Measuring Malmquist productivity index: A new approach based on double frontiers data envelopment analysis. *Math. Comput. Model.* **2011**, *54*, 2760–2771. [CrossRef]
- 34. Örkcü, H.H.; Baklikci, C.; Dogan, M.; Genc, A. An evaluation of the operational efficiency of Turkish airports using data envelopment analysis and the Malmquist productivity index: 2009–2014 case. *Transp. Policy* **2016**, *48*, 92–104. [CrossRef]
- 35. Ahn, Y.-H.; Min, H. Evaluating the multi-period operating efficiency of international airports using data envelopment analysis and the Malmquist productivity index. J. Air Transp. Manag. 2014, 39, 12–22. [CrossRef]
- 36. De Nicola, A.; Gitto, S.; Mancuso, P. Airport quality and productivity changes: A Malmquist index decomposition assessment. *Transp. Res. Part E* 2013, *58*, 67–75. [CrossRef]
- 37. Pires, H.M.; Fernandes, E. Malmquist financial efficiency analysis for airlines. *Transp. Res. Part E* 2012, *48*, 1049–1055. [CrossRef]
- Arabi, B.; Munisamy, S.; Emrouznejad, A.; Shadman, F. Power industry restructuring and eco-efficiency changes: A new slacks-based model in Malmquist-Luenberger Index measurement. *Energy Policy* 2014, 68, 132–145. [CrossRef]
- 39. Chang, D.-S.; Kuo, L.R.; Chen, Y. Industrial changes in corporate sustainability performance—An empirical overview using data envelopment analysis. *J. Clean. Prod.* **2013**, *56*, 147–155. [CrossRef]
- 40. Lin, E.Y.-Y.; Chen, P.-Y.; Chen, C.-C. Measuring green productivity of country: A generalized metafrontier Malmquist productivity index approach. *Energy* **2013**, *55*, 340–353.
- 41. Von Carlowitz, H.-C. Sylvicultura Oeconomica; J.F. Braun: Leipzig, Germany, 1713.
- 42. Meadows, D.H.; Meadows, D.L.; Randers, J.; Behrens, W.W. *The Limits to Growth*; Universe Books: New York, NY, USA, 1972.
- 43. Brundtlandt, G.M. *Our Common Future;* World Commission on Environment and Development: New York, NY, USA, 1988.
- 44. Kates, R.W.; Clark, W.C.; Corell, R.; Hall, M.; Jaeger, C.C.; Lowe, I.; McCarthy, J.J.; Schellnhuber, H.J.; Bolin, B.; Dickson, N.M.; et al. Sustainability Science. *Science* **2001**, *292*, 641–642. [CrossRef] [PubMed]
- 45. Holden, E.; Linnerud, K.; Banister, D. Sustainable development: Our Common Future revisited. *Glob. Environ. Chang.* **2014**, *26*, 130–139. [CrossRef]
- 46. Sneddon, C.; Howarth, R.B.; Norgaard, R.B. Sustainable development in a post-Brundtland world. *Ecol. Econ.* **2006**, *57*, 253–268. [CrossRef]
- 47. Robinson, J. Squaring the circle? Some thoughts on the idea of sustainable development. *Ecol. Econ.* **2004**, *48*, 369–384. [CrossRef]
- 48. Brockhaus, S.; Fawcett, S.E.; Knemeyer, A.M.; Fawcett, A.M. Motivations for environmental and social consciousness: Reevaluating the sustainability-based view. *J. Clean. Prod.* **2017**, *143*, 933–947. [CrossRef]
- Chang, R.-D.; Zuo, J.; Zhao, Z.-Y.; Zillante, G.; Gan, X.-L.; Soebarto, V. Evolving theories of sustainability and firms: History, future directions and implications for renewable energy research. *Renew. Sustain. Energy Rev.* 2017, 72, 48–56. [CrossRef]
- 50. European Commission. Reducing Emissions from Transport. Available online: http://ec.europa.eu/clima/policies/transport/index_en.htm (accessed on 9 February 2017).



- 51. European Environmental Agency (EEA). Greenhouse Gas Emissions from Transport. Available online: http://www.eea.europa.eu/data-and-maps/indicators/transport-emissions-of-greenhouse-gases/ transport-emissions-of-greenhouse-gases-6 (accessed on 9 February 2017).
- 52. Levallois, C. Can de-growth be considered a policy option? A historical note on Nicholas Georgescu-Roegen and the Club of Rome. *Ecol. Econ.* **2010**, *69*, 2271–2278. [CrossRef]
- 53. European Commission. *Roadmap to a Single European Transport. Area—Towards a Competitive and Resource Efficient Transport System;* COM(2011)144 Final; European Commission: Brussels, Belgium, 2011.
- 54. Oltean-Dumbrava, C.; Watts, G.; Miah, A. Towards a more sustainable surface transport infrastructure: A case study of applying multi criteria analysis techniques to assess the sustainability of transport noise reducing devices. *J. Clean. Prod.* **2016**, *112*, 2922–2934. [CrossRef]
- 55. Zheng, Y.; Liao, H.; Yang, X. Stochastic Pricing and Order Model with Transportation Mode Selection for Low-Carbon Retailers. *Sustainability* **2016**, *8*, 48. [CrossRef]
- 56. Postorino, M.N.; Mantecchini, L. A transport carbon footprint methodology to assess airport carbon emissions. *J. Air Transp. Manag.* **2014**, *37*, 76–86. [CrossRef]
- 57. Jevons, W.S. *The Coal Question: An Inquiry Concerning the Progress of the Nation, and the Probable Exhaustion of Our Coal Mines;* Macmillan: London, UK, 1865.
- 58. Klumpp, M. To Green or Not to Green: A Political, Economic and Social Analysis for the Past Failure of Green Logistics. *Sustainability* **2016**, *8*, 441. [CrossRef]
- 59. Sorrell, S. Jevons' Paradox revisited: The evidence for backfire from improved energy efficiency. *Energy Policy* **2009**, *37*, 1456–1469. [CrossRef]
- 60. Alcott, B. Jevons' paradox. Ecol. Econ. 2005, 54, 9-21. [CrossRef]
- 61. Deutsche Bahn AG. 2015 Integrated Report—More Quality, More Customers, More Success; Deutsche Bahn AG: Berlin, Germany, 2016.
- 62. Elkington, J. Cannibals with Forks—The Triple Bottom Line of 21st Century Business; Capstone: Oxford, UK, 1997.
- 63. Akhtar, P.; Tse, Y.K.; Khan, Z.; Rao-Nicholson, R. Data-driven and adaptive leadership contributing to sustainability: Global agri-food supply chains connected with emerging markets. *Int. J. Prod. Econ.* **2015**, *181*, 392–401. [CrossRef]
- 64. Formentini, M.; Taticchi, P. Corporate sustainability approaches and governance mechanisms in sustainable supply chain management. *J. Clean. Prod.* **2016**, *112*, 1920–1933. [CrossRef]
- 65. Hansen, T.; Coenen, L. The geography of sustainability transitions: Review, synthesis and reflections on an emergent research field. *Environ. Innov. Soc. Trans.* **2015**, *17*, 92–109. [CrossRef]
- 66. Alibeyg, A.; Contreras, I.; Fernández, E. Hub network design problems with profits. *Transp. Res. Part E* 2016, *96*, 40–59. [CrossRef]
- 67. Li, Y.; Chen, H.; Prins, C. Adaptive large neighborhood search for the pickup and delivery problem with time windows, profits, and reserved requests. *Eur. J. Oper. Res.* **2016**, *252*, 27–38. [CrossRef]
- 68. Shahraeeni, M.; Ahmed, S.; Malek, K.; Van Drimmelen, B.; Kjeang, E. Life cycle emissions and cost of transportation systems: Case study on diesel and natural gas for light duty trucks in municipal fleet operations. *J. Nat. Gas Sci. Eng.* **2015**, *24*, 26–34. [CrossRef]
- 69. Köllen, T. Reconceptualizing Profit-Orientation in Management: A Karmic View on 'Return on Investment' Calculations. *Philos. Manag.* **2016**, *15*, 7–20. [CrossRef]
- 70. Arena, M.; Conte, A.; Melancini, M. Linking environmental accounting to reward systems: The case of the Environmental Profit and Loss Account. *J. Clean. Prod.* **2015**, *108*, 625–636. [CrossRef]
- 71. Shipman, A.S.; Byrne, C.L.; Mumford, M.D. Leader vision formation and forecasting: The effects of forecasting extent, resources, and timeframe. *Leadersh. Q.* **2010**, *21*, 439–456. [CrossRef]
- Emerson, J. The Blended Value Proposition: Integrating Social and Financial Returns. *Calif. Manag. Rev.* 2003, 45, 35–51. [CrossRef]
- 73. Gambhir, A.; Tse, L.K.C.; Tong, D.; Martinez-Botas, R. Reducing China's road transport sector CO₂ emissions to 2050: Technologies, costs and decomposition analysis. *Appl. Energy* **2015**, *157*, 905–917. [CrossRef]
- 74. Mwangi, J.K.; Lee, W.-J.; Chang, Y.-C.; Chen, C.-Y.; Wang, L.-C. An overview: Energy saving and pollution reduction by using green fuel blends in diesel engines. *Appl. Energy* **2015**, *159*, 214–236. [CrossRef]
- 75. Walmsley, M.R.; Walmsley, T.G.; Atkins, M.J.; Kamp, P.J.J.; Neale, J.R.; Chand, A. Carbon Emissions Pinch Analysis for emissions reductions in the New Zealand transport sector through to 2050. *Energy* **2015**, *92*, 569–576. [CrossRef]



- 76. Gross, W.F.; Hayden, C.; Butz, C. About the impact of rising oil price on logistics networks and transportation greenhouse gas emission. *Logist. Res.* **2012**, *4*, 147–156. [CrossRef]
- 77. Rahman, S.; Serletis, A. The asymmetric effects of oil price and monetary policy shocks: A nonlinear VAR approach. *Energy Econ.* **2010**, *32*, 1460–1466. [CrossRef]
- 78. Wilford, A.L. Internal control reporting and accounting standards: A cross-country comparison. *J. Account. Public Policy* **2016**, *35*, 276–302. [CrossRef]
- 79. Jones, S.; Finley, A. Have IFRS made a difference to intra-country financial reporting diversity? *Br. Account. Rev.* 2011, 43, 22–38. [CrossRef]
- 80. OECD. OECD Data—Exchange Rates. Available online: https://data.oecd.org/conversion/exchange-rates. htm (accessed on 9 February 2017).
- 81. Wang, Z.; He, W. CO₂ emissions efficiency and marginal abatement costs of the regional transportation sectors in China. *Transp. Res. Part D* 2017, *50*, 83–97. [CrossRef]
- 82. Voytenko, Y.; McCormick, K.; Evans, J.; Schliwa, G. Urban living labs for sustainability and low carbon cities in Europe: Towards a research agenda. *J. Clean. Prod.* **2016**, *123*, 45–54. [CrossRef]
- 83. Chavez-Baeza, C.; Sheinbaum-Pardo, C. Sustainable passenger road transport scenarios to reduce fuel consumption, air pollutants and GHG (greenhouse gas) emissions in the Mexico City Metropolitan Area. *Energy* **2014**, *66*, 624–634. [CrossRef]
- 84. Manjunath, A.; Gross, G. Towards a meaningful metric for the quantification of GHG emissions of electric vehicles (EVs). *Energy Policy* **2017**, *102*, 423–429. [CrossRef]
- 85. Reichert, A.; Holz-Rau, C.; Scheiner, J. GHG emissions in daily travel and long-distance travel in Germany—Social and spatial correlates. *Transp. Res. Part D* 2016, *49*, 25–43. [CrossRef]
- 86. World Resources Institute/World Business Council for Sustainable Development. *The Greenhouse Gas. Protocol—A Corporate Accounting and Reporting Standard;* World Resources Institute/World Business Council for Sustainable Development: Washington, DC, USA; Geneva, Switzerland, 2004.
- Luthra, S.; Garg, D.; Haleem, A. An analysis of interactions among critical success factors to implement green supply chain management towards sustainability: An Indian perspective. *Resour. Policy* 2015, 46, 37–50. [CrossRef]
- 88. Lehtonen, M. The environmental-social interface of sustainable development: Capabilities, social capital, institutions. *Ecol. Econ.* **2004**, *49*, 199–214. [CrossRef]
- 89. Miller, J.W.; Saldanha, J.P. A new look at the longitudinal relationship between motor carrier financial performance and safety. *J. Bus. Logist.* **2016**, *37*, 284–306. [CrossRef]
- 90. Grund, C. Gender pay gaps among highly educated professionals—Compensation components do matter. *Labour Econ.* **2015**, *34*, 118–126. [CrossRef]
- 91. Buchanan, T. The Influence of Gender Role Attitudes on Perceptions of Women's Work Performance and the Importance of Fair Pay. *Sociol. Spectr.* **2014**, *34*, 203–221. [CrossRef]
- 92. Turker, D.; Altuntas, C. Sustainable supply chain management in the fast fashion industry: An analysis of corporate reports. *Eur. Manag. J.* 2014, *32*, 837–849. [CrossRef]
- 93. Panalpina World Transport (Holding) Ltd. 2015 Annual Report; Panalpina World Transport (Holding) Ltd.: Basel, Switzerland, 2016.
- 94. Al-Shaer, H.; Zaman, M. Board gender diversity and sustainability reporting quality. J. Contemp. Account. Econ. 2016, 12, 210–222. [CrossRef]
- 95. Perryman, A.A.; Fernando, G.D.; Tripathy, A. Do gender differences persist? An examination of gender diversity on firm performance, risk, and executive compensation. *J. Bus. Res.* **2016**, *69*, 579–586. [CrossRef]
- Wells, P. Diversity, scale and sustainability in human systems: Towards a research agenda. *J. Clean. Prod.* 2013, 47, 232–235. [CrossRef]
- 97. Panalpina World Transport (Holding) Ltd. 2012 Annual Report; Panalpina World Transport (Holding) Ltd.: Basel, Switzerland, 2013.
- 98. Kühne + Nagel. *Annual Report 2014—Global. Customer-Oriented. Innovative;* Kühne + Nagel: Schindellegi, Switzerland, 2015.
- 99. Kühne + Nagel. *Annual Report 2013—Continuity and Further Development;* Kühne + Nagel: Schindellegi, Switzerland, 2014.
- 100. Kühne + Nagel. *Annual Report 2012—People. Network. Objectives;* Kühne + Nagel: Schindellegi, Switzerland, 2013.



- 101. Kühne + Nagel. Annual Report 2011—Network and Product Portfolio—Kühne. + Nagel Accelerates Expansion; Kühne + Nagel: Schindellegi, Switzerland, 2012.
- 102. Kühne + Nagel. Annual Report 2010; Kühne + Nagel: Schindellegi, Switzerland, 2011.
- 103. Kühne + Nagel. Annual Report 2009; Kühne + Nagel: Schindellegi, Switzerland, 2010.
- 104. Kühne + Nagel. Annual Report 2008; Kühne + Nagel: Schindellegi, Switzerland, 2009.
- 105. Kühne + Nagel. Annual Report 2007; Kühne + Nagel: Schindellegi, Switzerland, 2008.
- 106. Kühne + Nagel. Annual Report 2006; Kühne + Nagel: Schindellegi, Switzerland, 2007.
- 107. Deutsche Bahn AG 2015. 2014 Integrated Report—Integrated Thinking. Sustainable Action. Long-Term Success; Deutsche Bahn AG: Berlin, Germany, 2015.
- 108. Deutsche Bahn AG. 2013 Integrated Report—DB2020: Our Compass, Even in Challenging Times; Deutsche Bahn AG: Berlin, Germany, 2014.
- 109. Deutsche Bahn AG. *Our Future: Sustainably Successful—Deutsche Bahn.* 2012 Annual Report; Deutsche Bahn AG: Berlin, Germany, 2013.
- 110. Deutsche Bahn AG. DB Mobility Logistics-2011 Annual Report; Deutsche Bahn AG: Berlin, Germany, 2012.
- 111. Deutsche Bahn AG. 2010 Annual Report; Deutsche Bahn AG: Berlin, Germany, 2011.
- 112. Deutsche Bahn AG. 2009 Annual Report; Deutsche Bahn AG: Berlin, Germany, 2010.
- 113. Deutsche Bahn AG. 2008 Annual Report; Deutsche Bahn AG: Berlin, Germany, 2009.
- 114. Deutsche Bahn AG. 2007 Annual Report; Deutsche Bahn AG: Berlin, Germany, 2008.
- 115. Deutsche Bahn AG. 2006 Annual Report; Deutsche Bahn AG: Berlin, Germany, 2007.
- 116. Deutsche Post DHL. When You Think of Logistics—2014 Annual Report; Deutsche Post AG: Bonn, Germany, 2015.
- 117. Deutsche Post DHL. Tailor Made—2013 Annual Report; Deutsche Post AG: Bonn, Germany, 2014.
- 118. Deutsche Post DHL. Pioneering Future Markets-2012 Annual Report; Deutsche Post AG: Bonn, Germany, 2013.
- 119. Deutsche Post DHL. Simply Grow—2011 Annual Report; Deutsche Post AG: Bonn, Germany, 2012.
- 120. Deutsche Post DHL. Being Your Choice—2010 Annual Report; Deutsche Post AG: Bonn, Germany, 2011.
- 121. Deutsche Post DHL. Showing Respect, Delivering Results—2009 Annual Report; Deutsche Post AG: Bonn, Germany, 2010.
- 122. Deutsche Post DHL. Delivering on the Future-2008 Annual Report; Deutsche Post AG: Bonn, Germany, 2009.
- 123. Deutsche Post DHL. Roadmap to Value—2007 Annual Report; Deutsche Post AG: Bonn, Germany, 2008.
- 124. Deutsche Post DHL. Building on Our Strategic Strengths—2006 Annual Report; Deutsche Post AG: Bonn, Germany, 2007.
- 125. DSV A/S. 2015 Annual Report; DSV A/S: Hedehusene, Denmark, 2016.
- 126. DSV A/S. 2014 Annual Report; DSV A/S: Hedehusene, Denmark, 2015.
- 127. DSV A/S. 2013 Annual Report; DSV A/S: Hedehusene, Denmark, 2014.
- 128. DSV A/S. 2012 Annual Report; DSV A/S: Hedehusene, Denmark, 2013.
- 129. DSV A/S. 2011 Annual Report; DSV A/S: Hedehusene, Denmark, 2012.
- 130. DSV A/S. 2010 Annual Report; DSV A/S: Hedehusene, Denmark, 2011.
- 131. DSV A/S. 2009 Annual Report; DSV A/S: Hedehusene, Denmark, 2010.
- 132. DSV A/S. 2008 Annual Report; DSV A/S: Hedehusene, Denmark, 2009.
- 133. DSV A/S. 2007 Annual Report; DSV A/S: Hedehusene, Denmark, 2008.
- 134. DSV A/S. 2006 Annual Report; DSV A/S: Hedehusene, Denmark, 2007.
- 135. Panalpina World Transport (Holding) Ltd. 2014 Annual Report; Panalpina World Transport (Holding) Ltd.: Basel, Switzerland, 2015.
- 136. Panalpina World Transport (Holding) Ltd. 2013 Annual Report; Panalpina World Transport (Holding) Ltd.: Basel, Switzerland, 2014.
- 137. Panalpina World Transport (Holding) Ltd. 2011 Annual Report; Panalpina World Transport (Holding) Ltd.: Basel, Switzerland, 2012.
- 138. Panalpina World Transport (Holding) Ltd. 2010 Annual Report—A Passion for Solutions; Panalpina World Transport (Holding) Ltd.: Basel, Switzerland, 2011.
- 139. Panalpina World Transport (Holding) Ltd. 2009 Annual Report—A Passion for Solutions; Panalpina World Transport (Holding) Ltd.: Basel, Switzerland, 2010.
- 140. Panalpina World Transport (Holding) Ltd. 2008 Annual Report—A Passion for Solutions; Panalpina World Transport (Holding) Ltd.: Basel, Switzerland, 2009.



- 141. Panalpina World Transport (Holding) Ltd. 2007 Annual Report—A Passion for Solutions; Panalpina World Transport (Holding) Ltd.: Basel, Switzerland, 2008.
- 142. Panalpina World Transport (Holding) Ltd. 2006 Annual Report—A Passion for Solutions; Panalpina World Transport (Holding) Ltd.: Basel, Switzerland, 2007.
- 143. Koopmans, T.C. Analysis of Production as an Efficient Combination of Activities. In Activity Analysis of Production and Allocation: Proceedings of a Conference; Koopmans, T.C., Ed.; Wiley: New York, NY, USA, 1951; pp. 33–97.
- 144. Debreu, G. The Coefficient of Resource Utilization. Econometrica 1951, 19, 273–292. [CrossRef]
- 145. Farrell, M.J. The Measurement of Productive Efficiency. J. R. Stat. Soc. Ser. A 1957, 120, 253–290. [CrossRef]
- 146. Diewert, W.E. Functional Forms for Profit and Transformation Functions. *J. Econ. Theory* **1973**, *6*, 284–316. [CrossRef]
- 147. Karsak, E.E.; Karadayi, M.A. Imprecise DEA framework for evaluating health care performance of districts. *Kybernetes* **2017**, *46*, 706–727. [CrossRef]
- 148. Bolli, T.; Olivares, M.; Bonaccorsi, A.; Daraio, C.; Aracil, A.D.; Lepori, B. The differential effects of competitive funding on the production frontier and the efficiency of universities. *Econ. Educ. Rev.* 2016, 52, 91–104. [CrossRef]
- Shwartz, M.; Burgess, J.F.; Zhu, J. A DEA based composite measure of quality and its associated data uncertainty interval for health care provider profiling and pay-for-performance. *Eur. J. Oper. Res.* 2016, 253, 489–502. [CrossRef]
- 150. Hollingsworth, B. Non-Parametric and Parametric Applications Measuring Efficiency in Health Care. *Health Care Manag. Sci.* **2003**, *6*, 203–218. [CrossRef] [PubMed]
- 151. Bessent, A.M.; Bessent, E.W.; Charnes, A.; Cooper, W.W.; Thorogood, N. Evaluation of educational program proposals by means of DEA. *Educ. Adm. Q.* **1983**, *19*, 82–107. [CrossRef]
- 152. Cooper, W.W.; Lawrence, M.S.; Tone, K. Data Envelopment Analysis—A Comprehensive Text with Models, Applications, References and DEA-Solver Software; Springer: New York, NY, USA, 2007.
- 153. Ding, L.; Yang, Q.; Sun, L.; Tong, J.; Wang, Y. Evaluation of the Capability of Personal Software Process based on Data Envelopment Analysis. In *Unifying the Software Process Spectrum*; Li, M., Boehm, B., Osterweil, L.J., Eds.; Springer: Berlin/Heidelberg, Germany, 2006; pp. 235–248.
- 154. Caves, D.W.; Christensen, L.R.; Diewert, W. The Economic Theory of Index Numbers and the Measurement of Input, Output, and Productivity. *Econometrica* **1982**, *50*, 1393–1414. [CrossRef]
- 155. Thanassoulis, E.; Shiraz, R.K.; Maniadakis, N. A cost Malmquist productivity index capturing group performance. *Eur. J. Oper. Res.* 2015, 241, 796–805. [CrossRef]
- 156. Emrouznejad, A.; Yang, G. A framework for measuring global Malmquist-Luenberger productivity index with CO₂ emissions on Chinese manufacturing industries. *Energy* **2016**, *115*, 840–856. [CrossRef]
- 157. Sueyoshi, T.; Goto, M. DEA environmental assessment in time horizon: Radial approach for Malmquist index measurement on petroleum companies. *Energy Econ.* **2015**, *51*, 329–345. [CrossRef]
- 158. Cook, W.D.; Zhu, J. Classifying inputs and outputs in data envelopment analysis. *Eur. J. Oper. Res.* 2007, 180, 692–699. [CrossRef]
- 159. Banker, R.D.; Morey, R.C. Efficiency analysis for exogenously fixed inputs and outputs. *Oper. Res.* **1985**, *34*, 513–521. [CrossRef]
- 160. Feng, C.; Chu, F.; Ding, J.; Bi, G.; Liang, L. Carbon Emissions Abatement (CEA) allocation and compensation schemes based on DEA. *Omega* **2015**, *53*, 78–89. [CrossRef]
- 161. Jain, R.K.; Natarajan, R. A DEA study of airlines in India. Asia Pac. Manag. Rev. 2015, 20, 285–292. [CrossRef]
- 162. DSV A/S. 2016 Annual Report; DSV A/S: Hedehusene, Denmark, 2017.
- 163. Dolinayova, A.; L'och, M.; Camaj, J. Liberalization of the railway freight market in the context of a sustainable transport system. *Transp. Res. Procedia* **2016**, *14*, 916–925. [CrossRef]
- 164. Bundesnetzagentur. *Railway Market. Analysis* 2015; Bundesnetzagentur für Elektrizität, Gas, Telekommunikation, Post und Eisenbahnen: Bonn, Germany, 2015.
- Nikitinas, V.; Dailydka, S. The models of management of railway companies in the European Union: Holding, the German Experience. *Procedia Eng.* 2016, 134, 80–88. [CrossRef]
- 166. European Commission. *Transport: Commission Urges Germany to Comply with EU Rules on the Separation of Accounts in Rail;* Press Release as of 20 June 2013; European Commission: Brussels, Belgium, 2013.



- Wang, Z.; Lu, M. An empirical study of direct rebound effect for road freight transport in China. *Appl. Energy* 2014, 133, 274–281. [CrossRef]
- 168. Winebrake, J.J.; Green, E.H.; Comer, B.; Corbett, J.J.; Froman, S. Estimating the direct rebound effect for on-road freight transportation. *Energy Policy* **2012**, *48*, 252–259. [CrossRef]
- 169. Matos, F.J.F.; Silva, F.J.F. The rebound effect on road freight transport: Empirical evidence from Portugal. *Energy Policy* **2011**, *39*, 2833–2841. [CrossRef]
- 170. Wieland, A.; Handfield, R.B.; Durach, C.F. Mapping the landscape of future research themes in supply chain management. *J. Bus. Logist.* **2016**, *37*, 205–212. [CrossRef]
- 171. Fawcett, S.E.; Waller, M.A. Supply chain game changers—Mega, nano, and virtual trends—And forces that impede supply chain design. *J. Bus. Logist.* **2014**, *35*, 157–164. [CrossRef]
- 172. Brockhaus, S.; Kersten, W.; Knemeyer, A.M. Where do we go from here? Progressing sustainability implementation efforts across supply chains. *J. Bus. Logist.* **2013**, *34*, 167–182. [CrossRef]
- 173. Simoes, P.; Marques, R.C. Influence of regulation on the productivity of waste utilities—What can we learn with the Portuguese experience? *Waste Manag.* **2012**, *32*, 1266–1275. [CrossRef] [PubMed]
- 174. Vlontzos, G.; Pardalos, P.M. Assess and prognosticate green house gas emissions from agricultural production of EU countries, by implementing, DEA Window analysis and artificial neural networks. *Renew. Sustain. Energy Rev.* 2017, 76, 155–162. [CrossRef]



© 2017 by the author. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (http://creativecommons.org/licenses/by/4.0/).



Reproduced with permission of copyright owner. Further reproduction prohibited without permission.

